Cyanine dyes in biophysical research: the photophysics of polymethine fluorescent dyes in biomolecular environments

Marcia Levitus^{1,2,3}* and Suman Ranjit^{1,2}

¹Center for Single Molecule Biophysics, Biodesign Institute, Arizona State University, Tempe, Arizona, USA

² Department of Chemistry and Biochemistry, Arizona State University, Tempe, Arizona, USA

³ Department of Physics, Arizona State University, Tempe, Arizona, USA

Abstract. The breakthroughs in single molecule spectroscopy of the last decade and the recent advances in super resolution microscopy have boosted the popularity of cyanine dyes in biophysical research. These applications have motivated the investigation of the reactions and relaxation processes that cyanines undergo in their electronically excited states. Studies show that the triplet state is a key intermediate in the photochemical reactions that limit the photostability of cyanine dyes. The removal of oxygen greatly reduces photobleaching, but induces rapid intensity fluctuations (blinking). The existence of non-fluorescent states lasting from milliseconds to seconds was early identified as a limitation in single-molecule spectroscopy and a potential source of artifacts. Recent studies demonstrate that a combination of oxidizing and reducing agents is the most efficient way of guaranteeing that the ground state is recovered rapidly and efficiently. Thiol-containing reducing agents have been identified as the source of long-lived dark states in some cyanines that can be photochemically switched back to the emissive state. The mechanism of this process is the reversible addition of the thiol-containing compound to a double bond in the polymethine chain resulting in a non-fluorescent molecule. This process can be reverted by irradiation at shorter wavelengths. Another mechanism that leads to non-fluorescent states in cyanine dyes is cis-trans isomerization from the singlet-excited state. This process, which competes with fluorescence, involves the rotation of one-half of the molecule with respect to the other with an efficiency that depends strongly on steric effects. The efficiency of fluorescence of most cyanine dyes has been shown to depend dramatically on their molecular environment within the biomolecule. For example, the fluorescence quantum yield of Cy3 linked covalently to DNA depends on the type of linkage used for attachment, DNA sequence and secondary structure. Cyanines linked to the DNA termini have been shown to be mostly stacked at the end of the helix, while cyanines linked to the DNA internally are believed to partially bind to the minor or major grooves. These interactions not only affect the photophysical properties of the probes but also create a large uncertainty in their orientation.

I. Introduction 124

2. Structure and nomenclature 126

3. Spectroscopic and photophysical properties 126

- 3.1. Spectral properties in solution 126
- 3.2. Photophysics of carbocyanines in solution 127

* Author for correspondence: M. Levitus, PO Box 875601, Tempe, AZ 85287-5601, USA. Tel.: (480) 727-8586; Fax: (480) 727-2378; Email: marcia.levitus@asu.edu

124 M. Levitus and S. Ranjit

- 3.3. Effects of substitutions |3|
- 3.4. Effects of conjugation to biomolecules |3|
- 3.5. Fluorescence quenching 133
- 3.6. Aggregation 134

4. Cyanine-DNA interactions 134

- 4.1. Modes of interaction and their effect on fluorescence properties 134
- 4.2. Significance in FRET spectroscopy 138

5. Photostability, photoblinking and photoswitching 139

- 5.1. Photostability 139
- 5.2. Photoblinking 140
- 5.3. Photoswitching 143
- 6. Concluding remarks 145
- 7. Acknowledgements 146
- 8. References 146

I. Introduction

The cyanine dyes are among the oldest and most investigated family of synthetic pigments. Polymethine cyanine dyes are extensively used as photosensitizers in silver halide photography (Sahyun et al. 1995; Steiger et al. 1980), as mode-locking compounds in laser technology (Ishchenko, 1994), and in photovoltaic and solar cells (Ehret et al. 2001; Ma et al. 2008). They have also been broadly used in the life sciences and other biologically related disciplines as optical probes of membrane potential (Waggoner, 1976), organelle stains (Koning et al. 1993; Terasaki et al. 1984), labels for neuron pathway tracing (Honig & Hume, 1989) and as probes for membrane structure and dynamics (Greenberg & Axelrod, 1993; Wu et al. 1977). Yet, their popularity has soared in the last two decades due to their applications as fluorescent labels for proteins and nucleic acids in single molecule and other demanding fluorescence microscopy applications. The indocarbocyanines commonly known as Cy3 and Cy5 (Fig. 1) became the fluorophores of choice during the early years of single-molecule spectroscopy primarily due to their remarkable photostability, large absorption cross sections and fluorescence efficiencies, compatibility with common lasers and single-photon counting detectors, and commercial availability as derivatives for covalent labeling of proteins and nucleic acids (Ha, 2001; Roy et al. 2008). Single-molecule methods are unique in that they provide insights into complex biological processes that are otherwise masked by the asynchronous behavior of the huge number of molecules present in the sample. Representative examples of the many exciting applications of singlemolecule spectroscopy that used cyanine dyes include the study of DNA helicases (Ha et al. 2002), DNA and RNA four-way junctions (Hohng et al. 2004; McKinney et al. 2004), kinesin motor proteins (Tomishige et al. 2006), ribozyme catalysis (Zhuang et al. 2000, 2002), the ATPase/ synthase motor (Diez et al. 2004; Yasuda et al. 2003), DNA replication (Luo et al. 2007), transcription (Kapanidis et al. 2005b, 2006) and translation (Blanchard et al. 2004; Kim et al. 2007). Most of these examples involve the use of two fluorescent probes, and take advantage of the strong distance-dependence of the Förster resonance energy transfer (FRET) phenomenon as a means to measure molecular distances in single biomolecules. The applications of single-molecule FRET in biochemistry have been reviewed elsewhere (Joo et al. 2008; Zhao & Rueda, 2009).



Fig. 1. Chemical structure of the cyanine dyes discussed in this work. Left: generic structure of a polymethine cyanine dye-containing alkylic substituents at both nitrogen atoms. The dotted line represents one of the heterocyclic moieties depicted in the middle. Right: succinimidyl ester derivatives of Cy3, Cy5 and Cy3B.

Photoblinking, the transient population of non-fluorescent states, is a major concern in singlemolecule spectroscopy because these 'off' states can last milliseconds to seconds and therefore interfere with the observation of dynamical processes in these timescales. Blinking does not manifest itself in experiments involving large number of molecules due to its stochastic nature, but can lead to apparent changes in FRET efficiency when a single molecule is observed. Interestingly, the existence of these otherwise detrimental dark states has enabled the recent applications of cyanine dyes as photoswitchable fluorescent probes in super-resolution imaging. One such example is a technique known as stochastic optical reconstruction microscopy (STORM), which relies on the use of individual fluorescent molecules such as Cy5, Cy5.5 and Cy7 that are switched 'on' and 'off' by lasers of different frequencies (Bates *et al.* 2007; Rust *et al.* 2006). The investigation of the nature and properties of transient dark states is also important to understand the mechanisms of photobleaching (the photochemical destruction of the fluorophore), since 'off' states are often important intermediates in these irreversible processes.

Understanding the photophysical properties of fluorescent dyes is not only important in singlemolecule research but also critical in any quantitative application of fluorescence spectroscopy. One such example is the use of FRET to determine distances in biomolecular systems. The relationship between fluorescence intensity (or lifetime) and inter-dye distance depends on the photophysical and spectral properties of the dyes, as well as their relative orientation. As we will discuss throughout this review, all these factors can potentially depend on the environment of the dyes within the macromolecule, the method used for covalent tethering, and specific interactions between the dyes and the biopolymer. Although this is true, in principle, for all fluorescent probes, as it will be discussed in detail later, these effects are particularly important for the short cyanines such as the popular dye known as Cy3.

This review focuses on the cyanine dyes that are used in biophysical research, usually as fluorescent probes to investigate the structure and dynamics of biopolymers. We will focus on the photophysical properties of the dyes that are relevant for the interpretation of quantitative fluorescence experiments other than the standard applications of cyanine dyes as DNA-staining agents. The applications of symmetric and asymmetric cyanine dyes as DNA-binding compounds for DNA detection and visualization have been reviewed elsewhere (Haugland, 1992; Hirons *et al.* 1994; Netzel *et al.* 1995; Rye *et al.* 1992). Section 2 contains a brief discussion on the structure of cyanine dyes and the nomenclature we will follow in this review. Section 3 focuses on the spectroscopic and photophysical properties of these dyes, including the effects of conjugation to biomolecules, quenching and aggregation. A large portion of these studies have been conducted in organic solvents (usually alcohols), and results in aqueous media are discussed whenever available. Section 4 focuses on a series of papers that have investigated the modes of cyanine–DNA interactions and their consequences in FRET spectroscopy. Finally, section 5 contains a discussion on photostability and photoblinking, and the recent applications of cyanine dyes as photoswitchable dyes.

2. Structure and nomenclature

In general terms, cyanine dyes consist of two nitrogen atoms linked by a conjugated polymethine chain containing an odd number of carbon atoms (Mishra et al. 2000). Polymethine cyanine dyes are not stable unless they are substituted with heterocyclic groups such as indole, quinoline, benzoxazole or benzothiazole at both ends of the chain (Fig. 1). The symmetric cyanines discussed in this review are named using the generic nomenclature $diXC_m(n)$, where n is the number of carbon atoms in the polymethine chain, *m* is the number of carbon atoms in the primary alkyl substituents attached to the nitrogens, and X is I, Q, O, or S for cyanines containing indole, quinoline, benzoxazole or benzothiazole heterocyclic groups, respectively. Alternatively, the acronyms DTCI, DTDCI and DTTCI have been used to designate the thiacarbocyanines with x=1, 2 and 3, respectively, DOCI, DODCI and DOTCI for the oxacarboyanines, and HICI, HIDCI and HITCI for the indocarbocyanines. In addition, the popular term 'Cy-dye' was introduced by Waggoner and co-workers to identify a series of indocarbocyanine dyes for fluorescent labeling of proteins and nucleic acids (Ernst et al. 1989; Southwick et al. 1990). The most popular fluorophores of this series, Cy3 and Cy5, are based on $diIC_2(3)$ and $diIC_2(5)$, respectively. Unfortunately, the term 'Cy-dye' has been widely used in the literature to designate different derivatives of these fluorophores, and therefore they do not represent unique chemical compounds. The main source of uncertainty when using this terminology is whether the phenyl rings are substituted with sulfonate moieties (Fig. 1). Sulfonates are typically present to increase solubility in aqueous buffers and minimize aggregation, so they are found in the commercially available succinimidyl ester and maleimide derivatives (GE Healthcare). In contrast, they are not present in the phosphoramidite derivatives used for solid-state synthesis of oligonucleotides (see Fig. 4a). Because cyanine-biomolecule interactions play a crucial role in determining the photophysical properties of the dye, the knowledge of the exact structure of the dye and the tether used for attachment are vital for the interpretation of any quantitative application involving Cy-dyes.

3. Spectroscopic and photophysical properties

3.1 Spectral properties in solution

Absorption in the visible region of the spectrum is determined by the existence of the delocalized π -electron system and therefore depends on the length of the polymethine chain. Figure 2 shows the absorption and emission spectra of diIC₂(3) and diIC₂(5) in methanol. The indocarbocyanine



Fig. 2. Absorption and emission spectra of diIC₂(3) (black) and diIC₂(5) (grey) in methanol (left, from Du *et al.* 1998) and the succinimidyl ester derivatives of Cy3 (black) and Cy5 (grey) in water (right, measured in the Levitus lab).

dilC₂(3) is characterized by an intense absorption band at 546 nm (in ethanol, $\varepsilon^{\max} = 1.33 \times 10^5/\text{cm/M}$) and a fluorescence emission spectrum with a maximum at 563 nm (Sims *et al.* 1974). The presence of heavy atoms at the end groups changes the effective length of the π -electron system and therefore has an impact in the absorption and emission maxima. For example, the thiacarbocyanine diSC₂(3) has absorption and emission maxima at 557 and 575 nm, respectively, whereas the oxacarbocyanine diOC₂(3) presents a significant shift to the blue ($\lambda_{abs}^{max} = 483$ nm, $\lambda_{em}^{max} = 499$ nm). The indodicarbocyanines (x=2, Fig. 1) and indotricarbocyanines (x=3) present intense absorption in the red and in the near infrared, respectively. The absorption and fluorescence spectral properties of diIC₂(5) and diIC₂(7) in ethanol are $\lambda_{abs}^{max} = 639$ nm, $\varepsilon^{max} = 2.40 \times 10^5/\text{cm/M}$ and $\lambda_{em}^{max} = 768$ nm for diIC₂(7) (Sims *et al.* 1974). All cyanine dyes exhibit small Stokes shifts and a small bathochromic displacement in absorption (red-shift) with decreasing solvent polarity (Åkesson *et al.* 1985; West & Geddes, 1964; Yu *et al.* 2002). Figure 2 shows the absorption and emission spectra of the dyes diIC₂(3) and diIC₂(5) in methanol (left) and the succinimidyl ester derivatives of Cy3 and Cy5 in water (right).

3.2 Photophysics of carbocyanines in solution

Cyanines can potentially form different isomers by rotation around the C–C bonds of the polymethine chain. In the ground state, polymethine dyes exist in the all-*trans* form unless they are sterically hindered (Kolesnikov & Mikhailenko, 1987; West *et al.* 1967; Wheatley, 1959). Thiacarbocyanines bearing bulky substituents in the polymethine chain have been shown to exist in a ground state equilibrium between the all-*trans* and the mono-*cis* isomer that is governed by solvent polarity (Khimenko *et al.* 1997; West *et al.* 1967). Isomerization from the excited state (photoisomerization) has been extensively studied and will be discussed in detail below because it represents the most efficient mechanism for excited state deactivation in fluid solution.

The photophysical behavior of polymethine cyanine dyes is usually described in terms of the potential energy surface depicted in Fig. 3, first proposed by Rulliere (1976). Following light absorption, the singlet-excited state deactivates by competing processes, the most efficient being fluorescence emission, internal conversion (ic) and rotation around a C–C bond of the



Fig. 3. Potential energy diagram for cyanine photoisomerization. The energies of the ground and first singlet excited states are represented as a function of torsion angle (θ). N represents the normal form (*trans* isomer), t the twisted state, and P the *cis* photoisomer. k_{ic} and k_{f} represent the internal conversion and radiative fluorescence rates, respectively.

polymethine chain. The relative efficiency of photoisomerization with respect to the other two processes depends on temperature, solvent viscosity and the presence of substituents that might create steric hindrance. Isomerization from the excited singlet state occurs via a nonspectroscopic partially twisted intermediate (t) which deactivates rapidly to the ground state hypersurface to yield the ground state photoisomer (P), or to return to the thermodynamically stable all-trans ground state (N) (Momicchioli et al. 1988; Murphy et al. 1994; Ponterini & Momicchioli, 1991). The photoisomer has been proposed to have a mono *is* conformation (Aramendia et al. 1994; Chibisov et al. 1996) and exhibits very low-fluorescence quantum yield (Dempster et al. 1972; Dipaolo et al. 1995; Duchowicz et al. 1990; Kuzmin & Darmanyan, 1978). Once formed, the photoisomer undergoes a thermal back-isomerization reaction to yield the thermodynamically stable all-*trans* isomer ($P \rightarrow N$). This process has been investigated extensively and has been found to be a first-order reaction with a rate that depends strongly on solvent viscosity, (Akesson et al. 1985; Aramendia et al. 1994; Chibisov et al. 1996; Korppitommola et al. 1991; Sundstrom & Gillbro, 1982; Waldeck & Fleming, 1981). In contrast, the influence of solvent polarity has been shown to be much less significant (Ponterini & Momicchioli, 1991; Sauerwein et al. 1992).

The excited singlet state of the shorter cyanines (n=1) in fluid solution is characterized by a short lifetime (τ_f) and low-fluorescence quantum yield (ϕ_f) due to very efficient rotation about the polymethine C–C bond (Åkesson *et al.* 1985; Korppitommola *et al.* 1991; Sibbett *et al.* 1981). The efficiency of fluorescence increases significantly when bond rotation is sterically hindered, as observed when the dyes are dissolved in highly viscous solvents (Åkesson *et al.* 1985; Sundstrom & Gillbro, 1982; Waldeck & Fleming, 1981) or bound to biomolecules (Brismar *et al.* 1995; Gruber *et al.* 2000; Harvey & Levitus, 2009). Rotation can also be eliminated altogether by chemical rigidization of the polymethine chain. Several rigidized trimethine cyanine dyes have been patented, but only the compound known as Cy3B (Fig. 1) is commercially available at present (Waggoner & Mujumdar, 2000).

The photophysical properties of $diIC_2(3)$ were investigated in fluid solutions as early as 1978, and are summarized in Table 1 together with the corresponding values for $diIC_2(5)$.

	$diIC_2(3)$	diIC ₂ (5)	
$\lambda_{\max}^{abs} (nm)$ $\varepsilon_{\max} (/cm/M)$ $\lambda_{\max}^{em} (nm)$ ϕ_{f} $\tau_{f} (ps)$ ϕ_{ISC} ϕ_{ic} ϕ_{Nt}	$546^{a} \\ 1.33 \times 10^{5 a} \\ 563^{a} \\ 0.042^{b} \\ 162^{d} \\ < 5 \times 10^{-3 c} \\ 0.06^{f} \\ 0.9^{h} $	$639^{a} 2.00 \times 10^{5 a} 664^{a} 0.21^{c} 980^{c} <3 \times 10^{-3 c} 0.11^{g} 0.7^{h} $	
$E_{ m Nt}$ (kJ/mol) $k_{ m PN}$ (/s) $E_{ m PN}$ (kJ/mol) Branching	19.7^{d} $10^{5 c}$ 45^{c} $2:1^{b}$	25^{i} $1 \cdot 1 \times 10^{3 i}$ 51^{i} $1 : 17^{c}$	

Table 1. Spectroscopic and photophysical parameters of $diIC_2(3)$ and $diIC_2(5)$ in ethanol (except when noted).

 λ_{\max}^{abs} : absorption maximum; ε_{\max} : extinction coefficient at λ_{\max}^{abs} ; λ_{\max}^{em} : fluorescence maximum; $\phi_{f:}$ fluorescence quantum yield; τ_f (ps): fluorescence lifetime; ϕ_{ISC} : quantum yield of inter-system crossing; ϕ_{ic} : quantum yield of internal conversion: ϕ_{Nt} : quantum yield of bond rotation (N \rightarrow t); E_{Nt} (kJ/mol): activation energy for the process N \rightarrow t; k_{PN} : rate constant for back-isomerization (P \rightarrow N) at room temperature; E_{PN} : activation energy for the process P \rightarrow N; branching: k_{tP}/k_{tN} .

^aSims et al. (1974).

^bdiIC₁(3) BF_4 (Chibisov *et al.* 1995).

^cChibisov et al. (1996).

^dÅkesson et al. (1985).

^eIn propanol (Kuzmin & Darmanyan, 1978).

^tEstimated as $1 - \phi_f$ (77 K) (Chibisov *et al.* 1995).

^gEstimated as $1 - \phi_f$ (-196 °C) (Chibisov *et al.* 1996).

^hEstimated as $1 - \phi_{\rm f} - \phi_{\rm ic} - \phi_{\rm ISC}$.

ⁱAramendia et al. (1994).

The quantum yield of fluorescence and intersystem crossing (ISC) of diIC₂(3) in propanol at room temperature were reported as $\phi_f = 0.028$ and $\phi_{ISC} < 5 \times 10^{-3}$, respectively (Kuzmin & Darmanyan, 1978). Rotation about a C-C bond is extremely efficient ($\phi_{Nt}=0.9$) and is responsible for the very short fluorescence lifetime of diIC₂(3) ($\tau_{\rm f}$ = 162 ps in ethanol) (Åkesson et al. 1985). The lifetime of fluorescence depends strongly on temperature and solvent viscosity because isomerization is an activated process that involves a large molecular motion. Åkesson et al. (1991) and Korppitommola et al. (1991) independently determined the activation energies of the N->t process in a series of primary alcohols of increasing viscosity from methanol ($E_{\rm Nt}$ = 11.7 kJ/mol), to dodecanol ($E_{\rm Nt}$ = 29.9 kJ/mol). A value of 19 kJ/mol was reported for the succinimidyl ester derivative of Cy3 (Cy3-SE) in a buffer solution (Sanborn et al. 2007), and is very similar to the one obtained for $diIC_2(3)$ in primary alcohols of similar viscosities (Akesson et al. 1991; Korppitommola et al. 1991). Bond rotation leads to the formation of the partially twisted intermediate, which deactivates to the cis or trans ground states with a branching ratio $(k_{tP}:k_{tN})$ of approximately 2:1 for diIC₁(3) in ethanol (Chibisov *et al.* 1995). The *cis* isomer reverts thermally to the more stable *trans* isomer with a first-order rate constant $k_{\rm PN} = 10^5$ /s (propanol and room temperature) and activation energy of E_{PN}=45 kJ/mol (Kuzmin & Darmanyan, 1978). The efficiency of internal conversion of diIC₁(3) was estimated as $\phi_{ic} = 0.06$ from the measurement of the fluorescence quantum yield at 77 K, when isomerization is completely suppressed (Chibisov *et al.* 1995). A value of $\phi_{ic} = 0.15$ was estimated for Cy3-SE in an aqueous buffer using similar arguments, and by comparison with the fluorescence quantum yield and

lifetime reported for the rigidized cyanine Cy3B (Cooper *et al.* 2004; Sanborn *et al.* 2007). The limited number of studies on the photophysical properties of the dye Cy3 show that, as expected, its behavior closely resembles that of the well-characterized dye diIC₂(3). The triplet state properties of Cy3-SE have been recently investigated in more detail due to the involvement of the triplet state in photophysical phenomena of interest in single-molecule spectroscopy and super-resolution imaging (e.g. photoblinking and photoswitching, see section 5). The quantum yield of triplet formation of Cy3-SE in argon-saturated methanol solution has been determined as $\phi_{\rm ISC} = 0.03$. The triplet state of the *trans* isomer shows absorption at 580 nm ($\varepsilon_{\rm T} = 40.770/M/cm$), and largely overlaps with the ground state absorption of the *cis* Cy3-SE isomer (Jia *et al.* 2007). Due to its rigid structure, the fluorescence quantum yield and lifetime of Cy3B are larger than the corresponding values for Cy3. Cooper *et al.* reported $\phi_{\rm f} = 0.67$ and $\tau_{\rm f} = 2.8$ ns in aqueous buffer, while Sanborn *et al.* reported $\phi_{\rm f} = 0.85$ and $\tau_{\rm f} = 2.7$ ns (Cooper *et al.* 2004; Sanborn *et al.* 2007). Despite its growing popularity in biophysical research, no other photophysical studies have been reported to date for Cy3B.

Longer lifetimes and moderate quantum yields of fluorescence are observed for the longer cyanines (x>1) in fluid solution. Kuzmin *et al.* reported $\phi_f = 0.24$ and $\phi_{ISC} < 5 \times 10^{-3}$ for $diIC_2(5)$ in propanol and indicated that the photoisomer formed by laser photolysis does not fluoresce (Kuzmin & Darmanyan, 1978). Chibisov *et al.* reported ϕ_f (24 °C)=0.21, ϕ_f $(-196 \ ^{\circ}C) = 0.89$, $\tau_{\rm f} (24 \ ^{\circ}C) = 0.98$ ns and $\phi_{\rm ISC} < 3 \times 10^{-3}$ for diIC₁(5) in ethanol (Chibisov *et al.* 1996). The quantum efficiency of internal conversion can be estimated from the results at low temperatures as $\phi_{ic} = 0.11$. The triplet lifetime ($\tau_T = 60 \ \mu s$) and maximum of triplet-triplet absorption ($\lambda_{\rm T}$ = 690 nm) were obtained using sensitized excitation (Chibisov *et al.* 1996, 2001), and similar values have been reported for the dye Cy5-SE (Huang et al. 2006). Although the longer cyanines have moderate fluorescence efficiencies, rotation about C-C bonds in the polymethine chain is still usually the most efficient deactivation pathway (the above values give $\phi_{Nt} \sim 0.7$). Yet, in contrast to what has been observed with the shorter cyanines, only a small fraction of the twisted intermediates decay to the *vis* isomer ($\phi_{tP} = 0.038$) (Chibisov *et al.* 1996). The activation energies for photoisomerization were reported for $diIC_2(5)$ in a series of normal alcohols of increasing viscosity (Aramendia et al. 1994; Chibisov et al. 1996). Values in ethanol are $E_{\rm Nt} = 25 \text{ kJ/mol}$ and $E_{\rm PN} = 51 \text{ kJ/mol}$.

The spectroscopic properties of cyanine dyes in aqueous buffers differ to some extent from those reported in normal alcohols. A 5-10 nm hypsochromic shift in absorption and emission spectra was reported for the sulfonate-substituted $diIC_2(3)$ and $diIC_2(5)$ in phosphate buffer with respect to ethanol (Mujumdar et al. 1993). The fluorescence quantum yield of these dyes is significantly lower in buffer than ethanol, even when the viscosities of these solvents are practically the same. Sauerwein *et al.* reported that the dynamics of photoisomerization of $diIC_1(3)$ correlates with the molecular mass of the solvent and not necessarily with viscosity, as it is seen when normal alcohols are used (Sauerwein et al. 1992). For instance, the lifetime of this compound in isomeric alcohols is nearly identical even when their viscosities are significantly different. In this work, the authors show a linear dependence between the lifetime of fluorescence of $diIC_1(3)$ and the molecular weight of the solvent, including water and several linear and branched alcohols. Therefore, the low-molecular mass of water might facilitate the transfer of momentum and therefore accelerate photoisomerization. In addition, there is no correlation between the measured fluorescence lifetime and solvent polarity. For instance, the lifetime of diIC₁(3) in ethanol, 2-pentanone and ethyl acetate are very similar, even when the dielectric constants of these solvents are markedly different (Sauerwein et al. 1992).

3.3 Effects of substitutions

The photophysical effects of substituents and other chemical modifications in polymethine cyanine dyes have been extensively investigated during the last few decades. Substitutions in the polymethine chain generally increase the rate of photoisomerization and decrease the efficiency of fluorescence. Chibisov *et al.* reported a sixfold increase in $\phi_{\rm NP}$ and a twofold decrease in $\phi_{\rm f}$ for diIC₁(5) substituted with a methyl group in the *meso*-position (center of the polymethine chain) (Chibisov et al. 1996). The bromine derivative also shows not only an enhancement in the isomerization rate but also a marked increase in intersystem crossing ($\phi_{\rm ISC} = 0.2$) due to heavyatom effects. Similar effects have been reported for alkyl-substituted thiacarbocyanines (Khimenko et al. 1997; Sibbett et al. 1981), which were also found to be less photostable than the unsubstituted compounds (Byers et al. 1976). For substituents at the aromatic rings, the fluorescence quantum yield increases significantly in the presence of strong electron-withdrawing groups such as NO_2 – and CF_3SO_2 –, and decreases in the presence of electron donating groups like MeO- (Mader et al. 2004; Murphy et al. 1994). Electron-withdrawing groups affect bond localization on the central C-C bonds, hindering the rotation and thus increasing the lifetime of the excited state (Murphy et al. 1994). Substituents at the indolic nitrogens affect the isomerization rates due to steric effects, as bond rotation is retarded by the solvent when the size of the rotating group increases. Akesson *et al.* investigated the photophysics of $diIC_2(3)$, $diIC_6(3)$ and diIC14(3) in normal alcohols, and observed a twofold increase in the fluorescence quantum yield of diIC₁₄(3) with respect to diIC₂(3) in methanol (Åkesson *et al.* 1991).

3.4 Effects of conjugation to biomolecules

The covalent attachment of fluorescent dyes to biomolecules is typically achieved by the reaction of a sulfhydryl group or a primary amine with the dye maleimide or N-Hydroxysuccinimide ester (NHS-ester), respectively (Fig. 4b, f) (Brinkley, 1992). The maleimide and NHS-esters of the sulfoindocarbocyanines Cy3 and Cy5 are commercialized by GE Healthcare and contain the reactive groups linked to one of the nitrogen atoms of the dye. The attachment of a bulky substituent such as a protein or a nucleic acid can potentially affect the photophysical properties of the dye considerably due to significant effects in the dynamics of *cis*-*trans* isomerization. These effects are much more pronounced for Cy3 than Cy5 due to the lower energy of activation for photoisomerization of the shorter cyanine in the unbound state.

Binding to biomolecules can dramatically lower the efficiency of photoisomerization, producing as a consequence a large increase in fluorescence quantum yield and lifetime. Brismar *et al.* reported a fivefold increase in fluorescence lifetime and a small bathochromic shift in fluorescence for Cy3 conjugated to immunoglobulin G (Brismar *et al.* 1995). Interestingly, Rasnik *et al.* reported significant variations in the fluorescence quantum yield of Cy3 covalently bound to a helicase depending on the amino acid used for labeling (Rasnik *et al.* 2004). In this work, the authors used eight different labeling sites for specific attachment of Cy3, and observed ϕ_f values in the range 0.27–0.48 for the helicase–DNA complex. The lowest values were measured for Cy3 bound to two sites in a flexible domain of the protein, while the highest values correspond to residues that are predicted to be closer to the DNA. This remarkable observation indicates that the ability of the dye to isomerize depends dramatically on local interactions between the dye and the protein or the DNA.

Similar variations have been reported for Cy3 bound covalently to DNA. The fluorescence quantum yield and lifetime of Cy3 on DNA depend on the type of linker used for the attachment



Fig. 4. Chemical structures of various Cy3-DNA conjugates (--a-e) and the generic Cy3-protein attachment (f) as discussed in the text. Analogous structures can be obtained with other Cy-dyes.

(Fig. 4), DNA sequence and secondary structure. Surprisingly, the fluorescence quantum yield and lifetime of Cy3 attached to the 3' or 5' terminus of an oligonucleotide can be significantly higher than the value measured after the oligo is annealed to its complementary strand to form a helical duplex (Massey *et al.* 2006; Sanborn *et al.* 2007). Variations in photophysical parameters have also been observed for Cy3 attachment at the DNA terminus (Fig. 4*a*) or internal attachment using a flexible linker (Fig. 4*b*) (Sabanayagam *et al.* 2005b; Sanborn *et al.* 2007). The disparity of fluorescence quantum yield values reported for Cy3 in different systems is summarized in Fig. 5, and is a consequence of the crucial effect of local interactions on the dynamics of photoisomerization. Sanborn *et al.* demonstrated that the fluorescence quantum yield of Cy3 on DNA correlates with the activation energy of bond rotation (N \rightarrow t) and inversely with the amount of *cis* isomer formed (Sanborn *et al.* 2007). Recent studies by Harvey *et al.* point toward the role of sequence-dependent Cy3–DNA interactions in determining the efficiency of isomerization (see below) (Harvey *et al.* 2009).

Changes in Cy5 fluorescence quantum yield and lifetime upon covalent attachment are not as dramatic as with Cy3. The attachment of Cy3 to the β subunit of F1-ATPase produces an eightfold increment in quantum yield, while the efficiency of fluorescence of Cy5 bound to the γ -subunit of the same protein is the same as the corresponding value for the Cy5-maleimide (Yasuda *et al.* 2003). Yet, some changes in Cy5 photophysics have been reported. The lifetime of Cy5 increases 1.5-fold when bound to immunoglobulin G (Gruber *et al.* 2000; Schobel *et al.* 1999), and a ~1.3-fold increase has been reported for Cy5 bound to the protein concanavalin A (Tolosa *et al.* 1997). Binding of Cy5 to proteins is usually accompanied by a small (~5 nm) bathochromic shift in fluorescence spectrum (Buschmann *et al.* 2003; Schobel *et al.* 1999). A large hypsochromic shift has been observed for large Cy5:protein labeling ratios and is likely due to



Fig. 5. Room-temperature fluorescence quantum yield of Cy3 in solution and covalently attached to various biopolymers. Cy3-SE PBS: Cy3 succinimidyl ester dissolved in phosphate-buffered saline solution (Sanborn *et al.* 2007); Cy3-SE glycerol: Cy3 succinimidyl ester dissolved in glycerol (Sanborn *et al.* 2007); +100 mM dAMP: Cy3 succinimidyl ester dissolved in phosphate-buffered saline solution containing 100 mM dAMP (Harvey & Levitus, 2009); ss poly(dA): Cy3 covalently linked to the 5' end of a 15 base-poly(dA) oligonucleotide (Harvey *et al.* 2009); ss poly(dT): Cy3 covalently linked to the 5' end of a 15 base-poly(dT) oligonucleotide (Harvey *et al.* 2009); ss 5' mixed and ds 5' mixed: Cy3 covalently linked to the 5' end of the oligonucleotide TTCTTCAGTTCAGCC and its corresponding double-stranded structure (Sanborn *et al.* 2007); ss int-mixed: Cy3 covalently linked to the sequence GGCTGAACTGAAGAG using the chemistry described in Fig. 3*b* at the site marked in bold face; helicase 1 and helicase 2: Cy3 covalently attached to a helicase using the chemistry shown in Fig. 3*f.* 'Helicase 1' corresponds to a site in the protein located in a flexible domain, and 'helicase 2' to residues that are predicted to be in a more restricted environment, closer to the DNA (Rasnik *et al.* 2004); F1-ATPase: Cy3 covalently attached to the β -subunit of F1-ATPase using the chemistry shown in figure 3*F* (Yasuda *et al.* 2003). The typical error in a fluorescence quantum yield determination is of the order of 10%.

the formation of aggregates (see below) (Gruber *et al.* 2000). Buschmann *et al.* investigated the properties of a series of red-absorbing fluorescent dyes and found that the binding of Cy5-biotin to streptavidin produces a red-shift in both absorption and emission spectra, and an increase in lifetime. The dynamics of isomerization was investigated by fluorescence correlation spectroscopy (FCS), and was found to be slower for the Cy5-biotin/streptavidin complex than for Cy5-biotin (Buschmann *et al.* 2003). Comparable results were reported by the same authors for the dye Alexa 647, which is structurally very similar to Cy5 (White *et al.* 2006), and for Cy5 attached to DNA and IgG (Widengren & Schwille, 2000).

3.5 Fluorescence quenching

Fluorescence quenching refers to any process that decreases the fluorescence intensity of the sample, and it is usually a consequence of collisional encounters between the fluorophore in the excited state and the quencher (dynamic quenching) or the formation of non-fluorescent ground state complexes (static quenching) (Valeur, 2001). Several fluorophores are efficiently quenched

by nucleobases and amino acids by both static and dynamic mechanisms. Marme et al. investigated the fluorescent quenching mechanisms of organic dyes by tryptophan and determined that the quenching of several red-absorbing dyes by tryptophan is dominated by the formation of non-fluorescent ground state complexes. However, quenching of the red-absorbing cyanines Cy5 and Alexa 647 by amino acids is negligible (Marme et al. 2003). Nucleobases can also quench the fluorescence of a large variety of fluorescent dyes by a combination of static and dynamic mechanisms. Photo-induced electron transfer has been identified as a common quenching mechanism, and its efficiency depends on the redox properties of the fluorescent dye. Rhodamine, Bodipy, oxazine and some coumarin fluorophores are efficiently quenched by guanosine, which has the highest electron donating ability among all nucleobases (Heinlein et al. 2003; Seidel et al. 1996; Torimura et al. 2001). Nucleobase reduction occurs with some coumarins, which are oxidized by photo-induced electron transfer interactions with deoxycytidine and thymidine (Seidel et al. 1996). The cyanines, Cy5 and Cy3, are not efficiently quenched by any of the nucleobases due to their lower electron accepting tendency (Torimura et al. 2001). Instead, the fluorescence quantum yield of cyanines is usually enhanced in the presence of nucleobases with respect to the values measured for the free dye in solution due to the effect of stacking interactions on the isomerization efficiency of the polymethine dye (Harvey & Levitus, 2009).

3.6 Aggregation

Cyanine dyes undergo aggregation to form dimers and more complex aggregates in aqueous solutions and in association with biological macromolecules. The aggregates of cyanines exhibit marked changes in the absorption band as compared to the monomers (Herz, 1977; Sims et al. 1974; West & Pearce, 1965). The dimerization of cyanines in water is usually accompanied by a reduction in the absorption of the main band and the appearance of a new maximum at shorter wavelengths (hypsochromic shift). A red-shifted peak (bathochromic shift) appears at higher concentrations due to higher-order aggregates, and is commonly referred to as the J-band (Jelley, 1936). H-aggregates are formed by molecules stacked in a parallel way (plane-to-plane), while J-aggregates are formed by head-to-tail arrangements (Harrison et al. 1996; Mishra et al. 2000). For the thiacarbocyanines, the tendency to aggregate has been shown to increase with the length of the polymethine chain (West & Pearce, 1965). Aggregated cyanine dyes can have longer excited singlet lifetimes but smaller fluorescence quantum yields than the monomers (Khairutdinov & Serpone, 1997). The dimers of several thiacarbocyanines have been shown to be non-fluorescent (Chibisov et al. 1999; Sims et al. 1974). The dicarbocyanine dye $diSC_2(5)$ was found to dimerize in the minor groove of DNA at alternating A-T or I-C sequences. Because the DNA structure limits the ability of additional dyes to stack onto the dimer, the aggregate propagates by a cooperative end-to-end mechanism (Armitage, 2005; Hannah & Armitage, 2004). The dye Cy5 was proposed to form non-fluorescent dimers when bound to proteins based on the observed hypsochromic shift in UV-vis absorption (Gruber et al. 2000).

4. Cyanine-DNA interactions

4.1 Modes of interaction and their effect on fluorescence properties

Small molecules can interact with DNA in a variety of ways, including intercalation, or binding to the minor or major grooves of the double helix. Intercalation and minor groove binding are the

most common modes of non-covalent binding of small molecules to DNA (Armitage, 2005). Intercalators are characterized by planar aromatic structures bearing a positive charge that generally exhibit a G-C preference, while minor groove binders are usually partially flexible and prefer A-T sites (Armitage, 2005; Haq, 2006). Both types of binding modes have been observed with cyanines, as the planar heterocycles favor intercalation while the polymethine carbon chain has enough flexibility to adapt to the curvature of the minor groove. As a consequence, small changes in dye structure or DNA sequence can cause a change from one binding mode to the other (Armitage, 2005). An extensive literature exists on the interactions of asymmetric cyanine dyes with DNA due to their applications to visualize and quantify DNA, and it will not be reviewed here (Haugland, 1992; Hirons et al. 1994; Netzel et al. 1995; Rye et al. 1992). The interactions of symmetrical cyanine dyes with DNA have been recently studied by Mikheikin et al. (2000). The cyanines $diQC_2(3)$, $diSC_2(3)$, $diOC_2(3)$ and $diIC_1(3)$ were shown to bind as monomers into the minor groove of the DNA with a slight preference toward A-T pairs. The binding constants correlate with the dye hydrophobicity as measured by the octanol/water partition coefficients, and follow the trend $diQC_2(3) > diSC_2(3) > diOC_2(3) > diOC_2(3)$. The complex has been proposed to be formed by binding of the cyanine into the DNA minor groove occupying the site of five base pairs. Binding of $diIC_1(3)$ to DNA is accompanied by a small bathochromic shift in absorption and a reduction in extinction coefficient of about 25%. At higher dye:DNA ratios, diQC₂(3) was shown to form a 2:1 complex involving a minor groove bound dimer (Mikheikin et al. 2000). The dye $diSC_2(5)$ was shown to form dimers that can extend into helical aggregates consisting of dimers aligned in an end-to-end fashion within the minor groove of DNA sequences containing alternating A-T or I-C residues (Armitage, 2005; Seifert et al. 1999).

In biophysical applications, fluorescent dyes are more often used as fluorescent labels attached to specific locations within the biomolecule. Recent experimental and theoretical studies of the dyes Cy3 and Cy5 attached covalently to DNA show that the dye interacts with the biopolymer regardless of the type of tether used for attachment. Lilley and co-workers characterized the structure of fluorescently labeled DNA by NMR, and showed that Cy3 and Cy5 covalently attached to the 5' terminus of DNA by a 3-carbon linker (Fig. 4a) forms a π -stacked complex with the terminal base pair as shown in Fig. 6 (Iqbal et al. 2008b; Norman et al. 2000). However, recent single-molecule and ensemble photophysical studies suggest that these interactions are dynamic, and the dyes likely exist in equilibrium between a bound and unbound state. Iqbal et al. studied the efficiency of fluorescence energy transfer between Cy3 and Cy5 terminally attached to the 5' termini of a DNA duplex using 3-carbon linkers (Iqbal et al. 2008a). The efficiency of FRET depends on distance and the relative orientation of the dyes (Dale et al. 1979). Because the dyes are mostly stacked on the ends of the helix, the observed FRET efficiency depends not only on the length of the helix, but also on the helical periodicity that determines the relative orientation between the donor and acceptor dyes. However, the observed modulation is less than that calculated for a fully rigid interaction between the fluorophores and the terminal bases. This is consistent with the results of Levitus and coworkers, who showed that although the NMR data indicate that most of the Cy3 is stacked on the end of the DNA, a fraction is unstacked and able to rotate rapidly around its linker (Sanborn et al. 2007).

The analysis of time-resolved fluorescence anisotropy data of Cy3 on DNA shows that the dye is mostly interacting with the DNA, not only when attached to the DNA terminus, but also when flexible linkers are used for covalent attachment (Sanborn *et al.* 2007). This is typically done



Fig. 6. A molecular graphics model of a DNA duplex with Cy3 (top) and Cy5 (bottom) fluorophores attached to the 5'-termini using the tether shown in Fig. 3*a* (courtesy of Professor D. Lilley). The model was generated from NMR structures of Cy3 and Cy5 attached to duplex DNA, and shows that the dyes are mostly stacked at the end of the helix (Iqbal *et al.* 2008b; Norman *et al.* 2000).

by incorporating a nucleobase with a reactive amine group during solid-state synthesis of the oligonucleotide, and subsequently coupling it with a succinimidyl ester derivative of the fluorophore. This approach leaves a rather long linker (often misleadingly called a 'six-carbon linker') between the dye and the nucleobase, which has enough length and flexibility to allow efficient dye–DNA interactions (Fig. 4*b*). Time-resolved fluorescence anisotropy is a technique based on the analysis of the polarization of the emitted light that provides a means by which the rotational mobility of the dye can be evaluated in the nanosecond timescale. Samples containing a Cy3 molecule attached covalently to DNA show two distinct times for rotational motion corresponding to fast rotations around the linker (180 ps) and slow motions that correspond to the overall tumbling of the DNA molecule and therefore indicate strong Cy3–DNA interactions. The timescale of this slow motion depends on the length and secondary structure of the DNA (4 ns for a 15 bp dsDNA and 2·5 ns for the single-stranded oligo) (Sanborn *et al.* 2007). The slow

timescale accounts for 82% of the decay for a 5'-modified ds DNA sample, and interestingly, for 88% of the decay for the single-stranded oligo of the same length. This shows that Cy3–DNA interactions are not only due to stacking at the end of the DNA as demonstrated by NMR, but also exist in the single-stranded oligo. Furthermore, 76% of the decay corresponds to slow tumbling motions in the sample containing the internal Cy3 with the six-carbon flexible linker, indicating that only 24% of the Cy3 molecules are in fact free to rotate around the tether. Molecular dynamics simulations of Cy5-labeled dsDNA containing a flexible linker show binding of the dye in two distinct locations of the major groove (Dolghih et al. 2007). Recent work by Singh et al. involving energy transfer between Cy5 and a gold nanoparticle provides further evidence of these interactions (Singh et al. 2009). These experiments are consistent with minor groove binding, and show that 37% of the dye is bound to the DNA when a six-carbon linker is used, while this fraction increases to 42% when a three-carbon linker is used instead. Remarkably, in contrast to the studies involving non-sulfonated Cy5 attached to the 3' terminus of a DNA duplex using 3-carbon linkers, the authors did not observe significant binding when the sulfonated dye is attached to the 5' terminus of the DNA via a six-carbon linker (Fig. 4e). This might be an effect of both the longer linker and the presence of the sulfonates on the Cy5 molecule.

Levitus and co-workers demonstrated that dye-DNA interactions result in dramatic variations in Cy3 fluorescence quantum yield and lifetime, which can potentially impact FRET and any other quantitative biophysical measurement using Cy3 fluorescence. The fluorescence of Cy3 is enhanced when the dye is dissolved in solutions containing nucleoside monophosphates (Harvey & Levitus, 2009). The effect is more pronounced for the purines (dAMP and dGMP) than the pyrimidines (dCMP and dTMP), and is consistent with a model in which Cy3-nucleoside π - π interactions decrease the efficiency of photoisomerization. The same experiments carried out with diIC₂(3) show a more pronounced enhancement of fluorescence quantum yield, pointing toward the role of the sulfonate groups present in Cy3 in inhibiting interactions with the DNA bases. The fluorescence quantum yield of Cy3 linked to the 5' end of oligonucleotides depends strongly on the DNA sequence. Harvey et al. reported the fluorescence properties of more than 20 Cy3-labeled oligos, and found dramatic variations in both the fluorescence quantum yield and lifetime (Harvey et al. 2009). While the maximum enhancement in solution is seen with the purine monophosphates, the fluorescence quantum yield of Cy3 attached to a poly(dA) oligo is among the lowest values measured in this work. This indicates that the nature of the interactions is radically different once the dye is covalently attached to the DNA, and suggests that interactions are likely reduced in oligos containing the dye next to stretches of purines due to their intrinsic rigidity. Therefore, DNA sequence affects the photophysical properties of Cy3 not only because of the different capabilities of the nucleobases to interact with the dye but also likely because DNA sequence determines the degree of flexibility of the polymer and therefore the likelihood that the dye will interact with the bases downstream.

Fluorophore–DNA interactions not only affect the photophysical properties of the dye but also can potentially alter the local structure of the biopolymer they are bound to. For instance, several fluorescent dyes have been found to stabilize the DNA duplex. The melting temperature of a 20 bp DNA increases up to 1.5 °C when fluorescent dyes are attached to the 5' terminus (Moreira *et al.* 2005). Interestingly, Cy3 and Cy5 produce the most pronounced effects, while fluorophores such as carboxytetramethylrhodamine, 6-carboxyfluorescein and hexachlorofluorescein have practically no impact on the melting properties of the duplex. These results are

consistent with the structural model of Lilley and co-workers, where the Cy-dyes stack on the terminal bases and virtually function as extra base pairs. The melting temperature of a 15 bp DNA modified internally with a Cy-dye via the common flexible alkane linker is 8-9 °C higher than that of the unlabelled duplex, indicating important perturbations of the DNA structure (Fegan *et al.* 2008).

4.2 Significance in FRET spectroscopy

Fluorophore–biomolecule interactions can potentially affect FRET measurements in many ways. On the one hand, interactions produce an important uncertainty in the kappa-square factor, which measures the relative orientation between the donor and the acceptor probes (Dale *et al.* 1979; van der Meer, 2002). This term can take values between 0 and 4 depending on the angles between the donor emission and the acceptor transition dipole moments, and equals 2/3 when both probes undergo unrestricted isotropic motion. The body of evidence presented above suggests that assuming unrestricted motion is generally not justified when Cy-dyes are used for FRET experiments, and therefore an uncertainty in the orientation factor will always exist. The use of flexible linkers not only does not guarantee rotational freedom, but also causes a large uncertainty in donor–acceptor distance. Molecular modeling shows that the flexibility of the six-carbon linker allows Cy5 to bind in two different locations in the major groove of the DNA, which result in an average donor–acceptor distance difference of 8 Å between the two conformers (Dolghih *et al.* 2007).

Some of the problems associated with the use of flexible linkers can be overcome by the use of rigid tethers. Fegan et al. used a rigid ethynyl linker on the C5 methyl group of thymidine to attach Cy3 and Cy5 to a DNA oligonucleotide (Fig. 4c) (Fegan et al. 2008). As opposed to what is observed when flexible linkers are used, the melting temperature of the duplexes modified with rigid labels was found to be the same as the unlabelled duplexes. This suggests that these rigid modifications do not perturb the structure of the DNA duplex. Ranjit et al. investigated the properties of Cy3 and Cy5 rigidly incorporated into the backbone of the DNA (Fig. 4d) (Ranjit et al. 2009). These modifications are commercially available and can be incorporated during solidstate synthesis using the corresponding Cy-dye phosphoramidite. Time-resolved fluorescence anisotropy experiments show that local motions are greatly restricted, minimizing the potential for dye-DNA interactions and the uncertainty in dye position that have been reported when flexible linkers are used. Interestingly, this type of modification allows some control over the orientation of the fluorophore. Samples in which the donor (Cy3) and acceptor (Cy5) are located on the same strand and separated by three helical turns present a high FRET efficiency, consistent with a kappa-square value close to the theoretical maximum encountered for co-linear dipoles ($\kappa^2 = 4$). The dependence of FRET with orientation was also verified for Cy3 and Cy5 attached to opposite termini of double-stranded DNA by Lilley and co-workers (Iqbal et al. 2008a). The authors took advantage of their previous structural studies that demonstrated that the dyes are mostly stacked at the ends of the DNA (Iqbal et al. 2008b; Norman et al. 2000), and therefore predicted that the relative orientation between the donor (Cy3) and acceptor (Cy5) molecules should depend on the length of the helix. The observed FRET efficiency does in fact depend on the length of the helix as well as on the helix periodicity (dsDNA versus RNA/ DNA hybrids), but the modulation is less than that calculated for completely rigid attachments because a fraction of the fluorophores exist in unstacked conformations for a fraction of the time.

The measurement of distances from FRET experiments also depends on other photophysical factors, including the donor fluorescence quantum yield, the acceptor extinction coefficient, and the donor fluorescence and acceptor absorption spectra. All these quantities are subject to variations depending on the local environment of the probes, and therefore they need to be evaluated for the particular system under study to avoid artifacts. All these factors are usually grouped under a quantity known as the Förster distance (R_0) , which measures the donoracceptor distance at which the efficiency of FRET is 0.5 (Valeur, 2001). Because this quantity is, in principle, a constant for a given donor-acceptor pair, researchers often rely on values published in the literature for the fluorophores of interest. This, however, can lead to important inaccuracies when cyanine dyes are used as fluorescent probes. For instance, reported R_0 values for the popular Cy3-Cy5 combination on biomolecules range from 47 to 65 Å when the orientation factor is assumed to be 2/3. This important disparity is a consequence of variations in the spectral properties of the dyes and the fluorescence quantum yield of Cy3 in the different molecular environments. Measurements on DNA systems range from 54 to 61 A (Iqbal et al. 2008a; Malicka et al. 2003; Murphy et al. 2004; Sabanayagam et al. 2005b), and measurements in a variety of proteins from 47 to 65 Å (Bastiaens & Jovin, 1996; Ishii et al. 1999; Lesoine et al. 2006; Rasnik *et al.* 2004; Yasuda *et al.* 2003). Therefore, it is clear that relying on a published R_0 can give rise to important errors even for biomolecules that can be regarded as similar at first sight (e.g. nucleic acids).

5. Photostability, photoblinking and photoswitching

5.1 Photostability

Photodegradation, commonly referred to as photobleaching, depletes the fluorophore concentration under prolonged irradiation and represents one of the most serious limitations in fluorescence microscopy. In single-molecule spectroscopy, the efficiency of photobleaching determines the total amount of photons that can be detected for a single fluorophore, and therefore imposes a limitation on the timescales that can be accessed by these measurements. Byers *et al.* investigated the photooxidation mechanisms of cyanine dyes, and reported quantum yields of photobleaching of less than 10^{-6} for a number of cyanines in methanol solution, including indo-, thia- and oxacarbocyanines (Byers *et al.* 1976). Indocarbocyanine dyes are more stable than thiaand oxacarbocyanines (Chen *et al.* 1999; Li *et al.* 1996; Sims *et al.* 1974), and increasing the polymethine chain length greatly reduces photostability (Waggoner *et al.* 2009). The triplet state is a key intermediate in the photodegradation of cyanines because it can participate in electron transfer reactions (Chibisov, 1977) and generate singlet oxygen by energy transfer. Singlet oxygen subsequently attacks the double bonds of the polymethine chain producing a compound that further reacts to form carbonyl products (Byers *et al.* 1976). Polyfluorination of diSC₂(5) has been shown to improve photostability significantly (Renikuntla *et al.* 2004).

The recognition of the role of oxygen in limiting the photostability of fluorescent dyes led to the use of oxygen removal enzymatic systems (oxygen scavengers) to increase observation times in fluorescence microscopy applications (Harada *et al.* 1990; Ishii *et al.* 1999). The most popular system makes use of a combination of the enzymes glucose oxidase and catalase in the presence of glucose. Glucose oxidase uses molecular oxygen to catalyze the oxidation of glucose into gluconic acid and hydrogen peroxide, which is subsequently decomposed into water and oxygen by the enzyme catalase (Benesch & Benesch, 1953; Englander *et al.* 1987). Overall, one molecule

of oxygen is removed per two molecules of glucose that are oxidized. The combination of this system and the reducing agent β -mercaptoethanol has been the gold standard in single-molecule fluorescence spectroscopy until recently (Ha, 2001). Glucose oxidase can be replaced by galactose oxidase in experiments using cell extracts containing endogenous glucokinase activities (Crawford *et al.* 2008). In either case, care must be taken because the accumulation of gluconic (or galactonic) acid produces a significant drop in pH, which can potentially affect the photophysical properties of the dye and alter protein stability (Sabanayagam *et al.* 2005a). An alternative enzymatic system that does not alter the pH of the buffer, and removes oxygen more efficiently, involves the use of protocatechuate dioxygenase (Aitken *et al.* 2008; Crawford *et al.* 2008; Patil & Ballou, 2000).

5.2 Photoblinking

The removal of oxygen, however, can induce rapid intensity fluctuations (blinking) because of the increased triplet state lifetime in anaerobic conditions. The triplet state lifetimes of cyanine dyes can increase significantly in the absence of oxygen, producing fluctuations in the millisecond timescale (English et al. 2000; Hubner et al. 2001; Kohn et al. 2002; Weston et al. 1999). Addition of compounds containing thiol groups to the oxygen scavenger system can suppress these fast fluctuations in Cy5, but can induce long-lived dark states (Heilemann et al. 2005; Rasnik et al. 2006; Sabanayagam et al. 2005a). For instance, Sabanayagam et al. characterized the blinking behavior of individual Cy5 molecules coupled covalently to DNA in the presence of an oxygen scavenger system containing β -mercaptoethanol (Sabanayagam *et al.* 2005a). The authors found that under red light illumination, the signal fluctuates between an 'on' state with an average lifetime of about 1 s and an 'off' state lasting tens of seconds. These long 'off' states are reversible, but can be easily mistaken by photobleaching events unless individual molecules are observed for prolonged periods of time. Interestingly, these transient dark states can be recovered by green illumination or by the presence of a donor, even under red illumination. These observations are the basis of the application of these dyes as 'on-off' switches, and will be discussed in detail below. Fluctuations in Cy5 fluorescence due to photophysical processes can be easily mistaken by fluctuations in FRET efficiency when this dye is used as an acceptor. The use of alternating laser excitation to excite the donor and acceptor directly is a useful approach to distinguish between dark acceptor states and states with an active acceptor but low FRET efficiency (Kapanidis et al. 2005a; Sabanayagam et al. 2005a). Recent publications in the field of nucleosome dynamics are a dramatic illustration of the potential artifacts that can arise from overlooking photophysical phenomena in single-molecule FRET measurements. Koopmans et al. demonstrated that most of the apparent FRET transitions reported previously by Leuba and co-workers were due to acceptor photophysics, and not to conformational dynamics as originally thought (Koopmans et al. 2007; Tomschik et al. 2005). The anti-correlated Cy3-Cy5 transitions observed in these experiments were initially interpreted as fluctuations in FRET efficiency due to changes in donor-acceptor distance in nucleosomes (Tomschik et al. 2005), but later proven to be due to Cy5 blinking by means of alternating laser excitation and the use of Trolox as an anti-blinking agent (Koopmans et al. 2007). Figure 7 shows fluorescence intensity traces from individual nucleosomes excited at 515 and 636 nm. When the donor is excited at 514 nm the donor and acceptor fluorescence intensities show anti-correlated changes (top panel) that translate into apparent FRET efficiency variations (bottom). However, Cy5 blinking becomes evident when the acceptor is excited directly using 633 nm excitation. In this case, the same



Fig. 7. Single-molecule FRET traces of individual nucleosomes labeled with a Cy3–Cy5 pair showing artifacts due to Cy5 blinking. The top panel shows the donor (green) and acceptor (red) intensities detected with 514 nm excitation. The middle panel shows the intensities measured with 636 nm excitation, providing evidence of acceptor blinking. The calculated FRET efficiency fluctuates between two values (bottom panel), which can be misinterpreted as nucleosome dynamics when only 514 nm excitation is used. Reproduced with permission from Koopmans *et al.* (2007). Copyright © 2007, Springer.

intensity fluctuations are observed, indicating that they are not related to distance-dependent changes in FRET efficiency but to acceptor photoblinking.

The quest to find the ideal anti-fading formula that does not induce blinking has been dominated by a rather empirical trial and error approach until recently. A variety of triplet quenchers and anti-oxidants have been investigated with varied success. Grunwell et al. tested several small-molecule compounds, including imidazole, β -mercaptoethanol, Trolox and propyl gallate. Trolox and propyl gallate were found to dramatically increase the photostability of Cy5 (Grunwell et al. 2001). A combination of propyl gallate and ascorbic acid has also proved to be effective with rhodamine 6 G (Widengren et al. 2007). Rasnik et al. investigated the use of Trolox to eliminate millisecond timescale fluctuations of Cy5 caused by removal of oxygen when enzymatic oxygen-scavenging systems are used. The authors concluded that compounds containing thiol groups (β -mercaptoethanol and L-glutathione) are efficient triplet state quenchers but cause slow blinking of Cy5, while antioxidants like propyl galate and ascorbic acid do not cause blinking but do not quench triplet states efficiently. In contrast, Trolox was the only compound to quench the triplet states of Cy5 and Alexa647 without inducing slow blinking (Rasnik et al. 2006). Dave et al. investigated the use of mercaptoethylamine, cyclooctatetraene, 4-nitrobenzyl alcohol and 1,4-diazabicyclo[2.2.2]octane in addition to some of the systems mentioned before, and found that their effectiveness depends strongly on the molecular environment of the dye. For instance, while Trolox improves the photostability of Cy5 in DNA systems, it actually reduces the 'on' time of the dye bound to tRNA molecules within the ribosome (Dave et al. 2009).

More recently, Tinnefeld and co-workers presented a more rational approach to the problem of improving photostability and reducing blinking. The authors reasoned that the key to minimize photobleaching and blinking is to recover reactive intermediates (triplet states and



Fig. 8. Single-molecule fluorescence traces of Cy5-labeled DNA immobilized on a coverslip in aqueous buffer containing an enzymatic oxygen scavenging system. (*a*) In the presence of 1 mM ascorbic acid and (*b*) in the presence of 1 mM ascorbic acid and 1 mM methyl viologen. In each case the bottom panel represents the data binned with 10 ms resolution, and the top trace a magnified portion binned with 1 ms resolution. Fluctuations due to blinking are evident in (*a*) and absent in (*b*). The autocorrelation decays in the inset provide further evidence of the presence and removal of fluorescence fluctuations due to blinking. Reproduced with permission from Vogelsang *et al.* (2008), Copyright Wiley-VCH Verlag GmbH & Co. KGaA.

photoionized species) quickly, and proposed the use of mixtures of reducing and oxidizing agents (Vogelsang *et al.* 2008). Triplet quenching by electron transfer produces radical anionic or cationic dye molecules, which need to be reoxidized or reduced to recover the singlet ground state. Ionized dyes can also be formed by other photo-induced electron transfer reactions, and represent important intermediates in photobleaching pathways. Therefore, the combination of both a reducing and an oxidizing agent allows the ground state to be recovered efficiently. This strategy has been elegantly demonstrated with several fluorophores, including the cyanines Cy5, Alexa 647 and Cy3B. Because of its general approach, the method is not limited to a particular chemical family of compounds, but is efficient for other fluorophores such as MR121 (an oxazine derivative), ATTO647N (a carbopyronine) and the rhodamines ATTO565 and Alexa 532. The photostability of all these dyes improves significantly in the presence of 1 mM methyl viologen (MV, and oxidant) and 1 mM ascorbic acid (AA, a reducing agent). For Cy5, the detected number of photons in PBS buffer after the removal of oxygen is below 1000 and increases to about 6×10^5 in the presence of these two compounds. Importantly, this increase in photostability is also accompanied by a dramatic reduction in blinking. Figure 8 shows

single-molecule fluorescence intensity traces of Cy5-labeled DNA in an aqueous buffer after oxygen removal in the presence of 1 mM AA (panel A) and 1 mM of both AA and MV (panel B). The top traces show an expanded view (1 ms resolution) of the bottom traces (10 ms resolution). The autocorrelation decays (right) demonstrate that the millisecond-timescale fluctuations observed in the presence of AA are only removed when MV is added to the buffer. Results with MV are only analogous to the results observed with AA (not shown). Interestingly, the same lab recently demonstrated that the effectiveness of the popular antiblinking and antibleaching reagent Trolox is due to the same principles outlined above (Cordes *et al.* 2009). A quinone derivative of Trolox is formed when the compound is slowly dissolved in aerated buffers, and acts as an oxidant in conjunction with Trolox, a reducing agent. Therefore, Trolox solutions contain the combination of reducing and oxidizing power required to quickly quench photo-induced cationic and anionic species to their singlet ground state.

Although applying a combination of reductant and oxidant systems has proved useful for a variety of fluorophores, dyes with large high electron affinities will require strong oxidants to reoxidize the reduced dye. This is in fact the case of ATTO 655 (an oxazine), which shows blinking in the presence of MV and AA in conditions where blinking is suppressed for most other fluorophores (Vogelsang *et al.* 2009). Similar results were recently obtained with a water-soluble perylene diimide (Cordes *et al.* 2010).

5.3 Photoswitching

The term 'photoswitching' refers to the reversible light-induced switching between a fluorescent ('on') and non-fluorescent ('off') state of a molecule. Reversible photoswitchable cyanines, including Cy5 and Alexa 647, are extensively used in super-resolution microscopy methods such as stochastic optical reconstruction microscopy (STORM). While the center of the image produced by a single molecule can be determined within 1.5 nm (Yildiz *et al.* 2003), the large number of molecules that are imaged in a biological sample makes this type of localization impossible. The key concept in STORM is to use photoswitchable probes to avoid collecting the fluorescence image of all molecules simultaneously. Because only a small subset of fluorescent molecules is activated at the time, the center of each individual molecule can be localized during a particular imaging cycle. This process is repeated numerous times until the position of all molecules is obtained so that the STORM fluorescence image can be reconstructed with nanometer accuracy (Bates *et al.* 2007; Rust *et al.* 2006).

Red emitting cyanine dyes such as Cy5, Alexa 647, Cy5·5 and Cy7 can act as photoswitches by themselves, or in the presence of an activator fluorophore nearby. Tinnefeld and co-workers demonstrated that the dye Cy5 can be recovered from its dark state by irradiation with light of 337, 488 or 532 nm (Heilemann *et al.* 2005). As a proof of principle, the authors showed that the fluorescence of a Cy5 molecule bound to DNA could be switched 'on' and 'off' about 100 times by applying alternating excitation at 633 and 488 nm. Similar findings were reported independently by Zhuang and co-workers using continuous 638 nm excitation to probe the fluorescence state of Cy5, and alternating 532 nm excitation to switch the molecule back to the fluorescent state (Bates *et al.* 2005). The rate constants of switching from 'on' to 'off' and 'off' to 'on' are linear with laser intensity, and the presence of a Cy3 molecule nearby causes a dramatic enhancement of conversion to the fluorescent state (Bates *et al.* 2005). The influence of green excitation and donor proximity in the blinking properties of Cy5 was also noted by Meller and co-workers (Sabanayagam *et al.* 2005a). These secondary fluorophores are known as activators,



Fig. 9. Photo-switchable probes using cyanine dyes. (*a*) Cy3 (activator) and Cy5, Cy5.5 and Cy7 as reporters. The 532 nm pulses that activate the reporter molecules are shown in green on the top panel. The dark yellow, red and brown lines show the fluorescence time traces of the reporter dye under 657 nm continuous excitation. (*b*) Cy5 activated by spectrally distinct activators. Bottom: Cy5 traces under continuous red laser excitation for probes using three different activators: Alexa 405 (magenta), Cy3 (green) and Cy2 (blue). The top panel shows the 405, 457 and 532 nm activation pulses that correspond to the absorption bands of the activators. Adapted from Bates *et al.* (2007). Reproduced with permission.

and allow the photoconversion to be carried out at much lower laser powers. Covalent heterodimers prepared from the Cy3-succinimidyl ester and the Cy5 hydrazide were also shown to behave as reversible photoswitches (Conley *et al.* 2008). Various dyes, including Alexa405, Cy2, TMR and Cy3, have been shown to act as activators when paired with Cy5 (Bates *et al.* 2007). In all cases, Cy5 was switched to the dark state with a red laser, and the activation back to the fluorescent state was achieved by excitation with a wavelength that corresponds to the absorption band of the activator chromophore (Fig. 9). This observation led to the development of multicolor STORM, where different populations of photoswitches containing different activators can be switched using lasers of different wavelengths. Two-dimensional, three-dimensional and multicolored images of nanoscopic structures in cells have been obtained using this technique (Bates *et al.* 2007; Huang *et al.* 2008a, b).

The restoration of absorption and fluorescence in all these cases depends critically on the absence of oxygen and the presence of a sulfur-containing triplet quencher-like β -mercaptoethanol (Bates et al. 2005; Heilemann et al. 2005). The rate constant for the transitions to the dark state does not depend on solvent viscosity, ruling out *cis-trans* isomerization as a possible mechanism for photoswitching. Different mechanisms have been speculated for the photoswitching mechanism of Cy5 and the nature of the dark state. Bates et al. suggested that the triplet state was an intermediate in the formation of the dark state based on the linear dependence of the 'off' rate with potassium iodide concentration (Bates et al. 2005). However, Heilemann et al. concluded that the triplet state is not involved in the formation of the switchable state based on the switching efficiency dependence on oxygen and triplet state quencher concentrations (Heilemann et al. 2005). Sauer and co-workers were the first to suggest that the switching mechanism involves the reversible conjugation of the thiol-containing compound to a double bond in the polymethine chain (Heilemann et al. 2009). This was recently verified by Zhuang and co-workers using a variety of chemical methods (Dempsey et al. 2009). The formation of a Cy5- β -mercaptoethanol adduct after red illumination is consistent with mass spectra analysis, and results suggests that the second carbon in the polymethine chain is the most plausible site of attachment. The disruption in the π -electron system is then responsible for the loss of fluorescence. Based on these findings, the most plausible mechanism for photoswitching involves a photochemical reaction between a double bond in the cyanine polymethine chain and a thiyl radical. The involvement of the radical species is consistent with the fact that the radical quencher isoascorbate reduces the switching rate to the dark state (Dempsey *et al.* 2009).

6. Concluding remarks

The popularity of cyanine dyes as fluorescent probes in biophysical research has steeply risen in the last decade thanks to advances in single-molecule methods and the development of superresolution microscopies. Technical developments increasingly allow researchers to carry out experiments with better sensitivity, improving the quantitative precision that can be achieved in these measurements. An interesting consequence of these technical advances is the growing interest in the biophysical community in characterizing and understanding the photophysical behavior of these probes. On one hand, the understanding of the nature and the dynamics of non-fluorescent states is crucial to characterize and find ways to improve photoblinking and photobleaching in measurements involving one or a few molecules. The role of buffer composition, and in particular the presence of thiol-containing molecules, was early identified as an important variable in the analysis of the blinking behavior of red-absorbing cyanines such as Cy5. Although most of the initial effort was directed toward finding conditions that would eliminate blinking in these molecules, it later became apparent that the existence of these long-lived dark states could be taken advantage of to construct photo-switchable probes. These probes are the basis of a novel super resolution imaging technique known as STORM.

In addition, the characterization of the photophysical behavior of fluorescent probes has proven to be critical for any quantitative study involving fluorescence, and in particular for the analysis of FRET experiments. There is extensive evidence that the fluorescence quantum yield and lifetime of cyanine dyes depends strongly on the molecular environment in which the probe is located, which determines the efficiency for *cis–trans* photoisomerization and therefore the lifetime of the singlet excited state. The photophysical properties of the dye Cy3 attached covalently to DNA have been shown to depend on the type of linker used for conjugation, DNA sequence and DNA secondary structure. Interactions with proteins are also important, as evidenced by studies that show that the efficiency of fluorescence of Cy3 linked to different cystein residues within the same protein can vary as much as 70%. Effects with Cy5 are not as significant, but still easily detectable. These factors have a considerable effect on the Förster distance of donor–acceptor FRET pairs involving cyanines, in particular when the shorter dye Cy3 is used as a donor, and should be evaluated carefully in any quantitative biophysical study.

The above discussion is somewhat unique to cyanine dyes due to their ability to isomerize from the first excited state. However, the photophysical properties of virtually all fluorophores are subject to some type of environment-related effect. The fluorescence quantum yield of fluorescein depends strongly on pH (Seybold *et al.* 1969), many coumarines, rhodamines, Alexa fluorophores, ATTO dyes and Bodipy derivaties are efficiently quenched by natural amino acids and nucleobases (Chen *et al.* 2010; Marme *et al.* 2003; Seidel *et al.* 1996; Torimura *et al.* 2001), and the spectral and fluorescent properties of many dyes depend on solvent polarity (Sackett & Wolff, 1987; Turner & Brand, 1968). Therefore, considering photophysical effects in biophysical research involving fluorescent probes is relevant in general, and sometimes even crucial.

7. Acknowledgements

The studies discussed in this review from the Levitus lab were supported by startup funds from Arizona State University and an NSF-CAREER grant. We thank Su Lin for help with the time-resolved spectroscopic measurements, and the following graduate and undergraduate students for their contributions to this research: Matthew Sanborn, Brian Connolly, Billie Jo Harvey, Kaushik Gurunathan, Claudia Perez and Priscilla Luna.

8. References

- AITKEN, C. E., MARSHALL, R. A. & PULGLISI, J. D. (2008). An oxygen scavenging system for improvement of dye stability in single-molecule fluorescence experiments. *Biophysical Journal* 94, 1826–1835.
- ÅKESSON, E., HAKKARAINEN, A., LAITINEN, E., HELENIUS, V., GILLBRO, T., KORPPITOMMOLA, J. & SUNDSTROM, V. (1991). Analysis of microviscosity and reaction coordinate concepts in isomerization dynamics described by Kramers' theory. *Journal of Chemical Physics* **95**, 6508– 6523.
- ÅKESSON, E., SUNDSTROM, V. & GILLBRO, T. (1985). Solvent-dependent barrier heights of excited-state photoisomerization reactions. *Chemical Physics Letters* 121, 513–522.
- ARAMENDIA, P. F., NEGRI, R. M. & SANROMAN, E. (1994). Temperature-dependence of fluorescence and photoisomerization in symmetrical carbocyanines – influence of medium viscosity and molecular-structure. *Journal of Physical Chemistry* **98**, 3165–3173.
- ARMITAGE, B. A. (2005). Cyanine dye–DNA interactions: intercalation, groove binding, and aggregation. DNA Binders and Related Subjects 253, 55–76.
- BASTIAENS, P. I. H. & JOVIN, T. M. (1996). Microspectroscopic imaging tracks the intracellular processing of a signal transduction protein: Fluorescent-labeled protein kinase c beta i. *Proceedings of the National Academy* of Sciences of the United States of America 93, 8407–8412.
- BATES, M., BLOSSER, T. R. & ZHUANG, X. W. (2005). Shortrange spectroscopic ruler based on a single-molecule optical switch. *Physical Review Letters* 94, 108101.
- BATES, M., HUANG, B., DEMPSEY, G. T. & ZHUANG, X. W. (2007). Multicolor super-resolution imaging with photoswitchable fluorescent probes. *Science* **317**, 1749–1753.
- BENESCH, R. E. & BENESCH, R. (1953). Enzymatic removal of oxygen for polarography and related methods. *Science* 118, 447–448.
- BLANCHARD, S. C., GONZALEZ, R. L., KIM, H. D., CHU, S. & PUGLISI, J. D. (2004). tRNA selection and kinetic proofreading in translation. *Nature Structural & Molecular Biology* **11**, 1008–1014.
- BRINKLEY, M. (1992). A brief survey of methods for preparing protein conjugates with dyes, haptens, and crosslinking reagents. *Bioconjugate Chemistry* 3, 2–13.

- BRISMAR, H., TREPTE, O. & ULEHAKE, B. (1995). Spectra and fluorescence lifetimes of lissamine rhodamine, tetramethylrhodamine isothiocyanate, texas red, and cyanine-3.18 fluorophores – influences of some environmental-factors recorded with a confocal laserscanning microscope. *Journal of Histochemistry and Cytochemistry* 43, 699–707.
- BUSCHMANN, V., WESTON, K. D. & SAUER, M. (2003). Spectroscopic study and evaluation of red-absorbing fluorescent dyes. *Bioconjugate Chemistry* 14, 195–204.
- BYERS, G. W., GROSS, S. & HENRICHS, P. M. (1976). Direct and sensitized photooxidation of cyanine dyes. *Photochemistry and Photobiology* 23, 37–43.
- CHEN, H., AHSAN, S. S., SANTIAGO-BERRIOS, M. E. B., ABRUÑA, H. D. & WEBB, W. W. (2010). Mechanisms of quenching of alexa fluorophores by natural amino acids. *Journal of the American Chemical Society* **132**, 7244–7245.
- CHEN, P., SUN, S. Q., HU, Y. F., QIAN, Z. G. & ZHENG, D. S. (1999). Structure and solvent effect on the photostability of indolenine cyanine dyes. *Dyes and Pigments* 41, 227–231.
- CHIBISOV, A. K. (1977). Triplet-states of cyanine dyes and reactions of electron-transfer with their participation. *Journal of Photochemistry* **6**, 199–214.
- CHIBISOV, A. K., SHVEDOV, S. V. & GORNER, H. (2001). Photosensitized processes in dicarbocyanine dyes induced by energy transfer: Delayed fluorescence, *trans-*> cis isomerization and electron transfer. Journal of Photochemistry and Photobiology A–Chemistry 141, 39–45.
- CHIBISOV, A. K., ZAKHAROVA, G. V. & GORNER, H. (1996). Effects of substituents in the polymethine chain on the photoprocesses in indodicarbocyanine dyes. *Journal of the Chemical Society–Faraday Transactions* **92**, 4917–4925.
- CHIBISOV, A. K., ZAKHAROVA, G. V. & GORNER, H. (1999). Photoprocesses in dimers of thiacarbocyanines. *Physical Chemistry Chemical Physics* 1, 1455–1460.
- CHIBISOV, A. K., ZAKHAROVA, G. V., GORNER, H., SOGULYAEV, Y. A., MUSHKALO, I. L. & TOLMACHEV, A. I. (1995). Photorelaxation processes in covalently-linked indocarbocyanine and thiacarbocyanine dyes. *Journal of Physical Chemistry* **99**, 886–893.
- CONLEY, N. R., BITEEN, J. S. & MOERNER, W. E. (2008). Cy3-Cy5 covalent heterodimers for single-molecule

photoswitching. Journal of Physical Chemistry B 112, 11878–11880.

- COOPER, M., EBNER, A., BRIGGS, M., BURROWS, M., GARDNER, N., RICHARDSON, R. & WEST, R. (2004). Cy3B[®] improving the performance of cyanine dyes. *Journal of Fluorescence* **14**, 145–150.
- CORDES, T., VOGELSANG, J. & TINNEFELD, P. (2009). On the mechanism of trolox as antiblinking and antibleaching reagent. *Journal of the American Chemical Society* 131, 5018–5019.
- CORDES, T., VOGELSANG, J., ANAYA, M., SPAGNUOLO, C., GIETL, A., SUMMERER, W., HERRMANN, A., MULLEN, K. & TINNEFELD, P. (2010). Single-molecule redox blinking of perylene diimide derivatives in water. *Journal of the American Chemical Society* **132**, 2404–2409.
- CRAWFORD, D. J., HOSKINS, A. A., FRIEDMAN, L. J., GELLES, J. & MOORE, M. J. (2008). Visualizing the splicing of single pre-mRNA molecules in whole cell extract. *RNA* 14, 170–179.
- DALE, R. E., EISINGER, J. & BLUMBERG, W. E. (1979). Orientational freedom of molecular probes – orientation factor in intra-molecular energy-transfer. *Biophysical Journal* 26, 161–193.
- DAVE, R., TERRY, D. S., MUNRO, J. B. & BLANCHARD, S. C. (2009). Mitigating unwanted photophysical processes for improved single-molecule fluorescence imaging. *Biophysical Journal* 96, 2371–2381.
- DEMPSEY, G. T., BATES, M., KOWTONIUK, W. E., LIU, D. R., TSIEN, R. Y. & ZHUANG, X. W. (2009). Photoswitching mechanism of cyanine dyes. *Journal of the American Chemical Society* 131, 18192.
- DEMPSTER, D. N., THOMPSON, G. F., MORROW, T. & RANKIN, R. (1972). Photochemical characteristics of cyanine dyes .1. 3,3'-diethyloxadicarbocyanine iodide. *Journal of the Chemical Society-Faraday Transactions II* 68, 1479.
- DIEZ, M., ZIMMERMANN, B., BORSCH, M., KONIG, M., SCHWEINBERGER, E., STEIGMILLER, S., REUTER, R., FELEKYAN, S., KUDRYAVTSEV, V., SEIDEL, C. A. M. & GRABER, P. (2004). Proton-powered subunit rotation in single membrane-bound F0F1-ATP synthase. *Nature Structural and Molecular Biology* **11**, 135–141.
- DIPAOLO, R. E., SCAFFARDI, L. B., DUCHOWICZ, R. & BILMES, G. M. (1995). Photoisomerization dynamics and spectroscopy of the polymethine dye dtci. *Journal of Physical Chemistry* **99**, 13796–13799.
- DOLGHIH, E., ROITBERG, A. E. & KRAUSE, J. L. (2007). Fluorescence resonance energy transfer in dye-labeled DNA. *Journal of Photochemistry and Photobiology A–Chemistry* 190, 321–327.
- DU, H., FUH, R. C. A., LI, J. Z., CORKAN, L. A. & LINDSEY, J. S. (1998). Photochemcad: a computer-aided design and research tool in photochemistry. *Photochemistry and Photobiology* 68, 141–142.
- Duchowicz, R., Scaffardi, L. & Tocho, J. O. (1990). Relaxation processes of singlet excited-state of

3,3'-diethyloxadicarbocyanine iodide (DODCI) photoisomer. *Chemical Physics Letters* **170**, 497–501.

- EHRET, A., STUHL, L. & SPITLER, M. T. (2001). Spectral sensitization of TiO₂ nanocrystalline electrodes with aggregated cyanine dyes. *Journal of Physical Chemistry B* 105, 9960–9965.
- ENGLANDER, S. W., CALHOUN, D. B. & ENGLANDER, J. J. (1987). Biochemistry without oxygen. *Analytical Biochemistry* 161, 300–306.
- ENGLISH, D. S., FURUBE, A. & BARBARA, P. F. (2000). Single-molecule spectroscopy in oxygen-depleted polymer films. *Chemical Physics Letters* 324, 15–19.
- ERNST, L. A., GUPTA, R. K., MUJUMDAR, R. B. & WAGGONER, A. S. (1989). Cyanine dye labeling reagents for sulfhydryl-groups. *Cytometry* **10**, 3–10.
- FEGAN, A., SHIRUDE, P. S. & BALASUBRAMANIAN, S. (2008). Rigid cyanine dye nucleic acid labels. *Chemical Communications*, 2004–2006.
- GREENBERG, M. L. & AXELROD, D. (1993). Anomalously slow mobility of fluorescent lipid probes in the plasmamembrane of the yeast saccharomyces-cerevisiae. *Journal of Membrane Biology* 131, 115–127.
- GRUBER, H. J., HAHN, C. D., KADA, G., RIENER, C. K., HARMS, G. S., AHRER, W., DAX, T. G. & KNAUS, H. G. (2000). Anomalous fluorescence enhancement of Cy3 and Cy3.5 versus anomalous fluorescence loss of Cy5 and Cy7 upon covalent linking to IgG and noncovalent binding to avidin. *Bioconjugate Chemistry* **11**, 696–704.
- GRUNWELL, J. R., GLASS, J. L., LACOSTE, T. D., DENIZ, A. A., CHEMIA, D. S. & SCHULTZ, P. G. (2001). Monitoring the conformational fluctuations of DNA hairpins using single-pair fluorescence resonance energy transfer. *Journal of the American Chemical Society* **123**, 4295–4303.
- HA, T. (2001). Single-molecule fluorescence resonance energy transfer. *Methods* 25, 78–86.
- HA, T., RASNIK, I., CHENG, W., BABCOCK, H. P., GAUSS, G. H., LOHMAN, T. M. & CHU, S. (2002). Initiation and re-initiation of DNA unwinding by the *Escherichia coli* rep helicase. *Nature* **419**, 638–641.
- HANNAH, K. C. & ARMITAGE, B. A. (2004). DNAtemplated assembly of helical cyanine dye aggregates: a supramolecular chain polymerization. *Accounts of Chemical Research* 37, 845–853.
- HAQ, I. (2006). Reversible small molecule–nucleic acid interactions. In *Nucleic acids in chemistry and biology* (eds. G. M. Blackburn, M. J. Gait, D. Loakes & D. M. Williams), pp. 341–379. Cambridge: Royal Society of Chemistry.
- HARADA, Y., SAKURADA, K., AOKI, T., THOMAS, D. D. & YANAGIDA, T. (1990). Mechanochemical coupling in actomyosin energy transduction studied by in vitro movement assay. *Journal of Molecular Biology* 216, 49–68.
- HARRISON, W. J., MATEER, D. L. & TIDDY, G. J. T. (1996). Liquid-crystalline J-aggregates formed by aqueous ionic cyanine dyes. *Journal of Physical Chemistry* 100, 2310–2321.

- HARVEY, B. J. & LEVITUS, M. (2009). Nucleobase-specific enhancement of Cy3 fluorescence. *Journal of Fluorescence* 19, 443–448.
- HARVEY, B. J., PEREZ, C. & LEVITUS, M. (2009). DNA sequence-dependent enhancement of Cy3 fluorescence. *Photochemical and Photobiological Sciences* 8, 1105–1110.
- HAUGLAND, R. P. (1992). Nucleic acid detection and analysis. In *Molecular Probes Handbook of Fluorescent Probes* and Research Chemicals. Eugene, OR: Molecular Probes.
- HEILEMANN, M., DEDECKER, P., HOFKENS, J. & SAUER, M. (2009). Photoswitches: Key molecules for subdiffraction-resolution fluorescence imaging and molecular quantification. *Laser and Photonics Reviews* 3, 180–202.
- HEILEMANN, M., MARGEAT, E., KASPER, R., SAUER, M. & TINNEFELD, P. (2005). Carbocyanine dyes as efficient reversible single-molecule optical switch. *Journal of the American Chemical Society* **127**, 3801–3806.
- HEINLEIN, T., KNEMEYER, J. P., PIESTERT, O. & SAUER, M. (2003). Photoinduced electron transfer between fluorescent dyes and guanosine residues in DNA-hairpins. *Journal of Physical Chemistry B* 107, 7957–7964.
- HERZ, A. H. (1977). Aggregation of sensitizing dyes in solution and their adsorption onto silver-halides. *Advances in Colloid and Interface Science* 8, 237–298.
- HIRONS, G. T., FAWCETT, J. J. & CRISSMAN, H. A. (1994). TOTO and YOYO – new very bright fluorochromes for DNA content analyses by flow-cytometry. *Cytometry* 15, 129–140.
- HOHNG, S., WILSON, T. J., TAN, E., CLEGG, R. M., LILLEY, D. M. J. & HA, T. J. (2004). Conformational flexibility of four-way junctions in RNA. *Journal of Molecular Biology* 336, 69–79.
- HONIG, M. G. & HUME, R. I. (1989). Dil and DiO – versatile fluorescent dyes for neuronal labeling and pathway tracing. *Trends in Neurosciences* 12, 333–341.
- HUANG, B., JONES, S. A., BRANDENBURG, B. & ZHUANG, X. W. (2008a). Whole-cell 3D storm reveals interactions between cellular structures with nanometer-scale resolution. *Nature Methods* 5, 1047–1052.
- HUANG, B., WANG, W. Q., BATES, M. & ZHUANG, X. W. (2008b). Three-dimensional super-resolution imaging by stochastic optical reconstruction microscopy. *Science* 319, 810–813.
- HUANG, Z. X., JI, D. M., WANG, S. F., XIA, A. D., KOBERLING, F., PATTING, M. & ERDMANN, R. (2006). Spectral identification of specific photophysics of Cy5 by means of ensemble and single molecule measurements. *Journal of Physical Chemistry A* **110**, 45–50.
- HUBNER, C. G., RENN, A., RENGE, I. & WILD, U. P. (2001). Direct observation of the triplet lifetime quenching of single dye molecules by molecular oxygen. *Journal of Chemical Physics* 115, 9619–9622.
- IQBAL, A., ARSLAN, S., OKUMUS, B., WILSON, T. J., GIRAUD, G., NORMAN, D. G., HA, T. & LILLEY, D. M. J. (2008a). Orientation dependence in fluorescent energy transfer between Cy3 and Cy5 terminally attached to

double-stranded nucleic acids. Proceedings of the National Academy of Sciences of the United States of America 105, 11176–11181.

- IQBAL, A., WANG, L., THOMPSON, K. C., LILLEY, D. M. J. & NORMAN, D. G. (2008b). The structure of cyanine 5 terminally attached to double-stranded DNA: implications for fret studies. *Biochemistry* 47, 7857–7862.
- ISHCHENKO, A. A. (1994). Laser media based on polymethine dyes. *Kvantovaya Elektronika* 21, 513–534.
- ISHII, Y., YOSHIDA, T., FUNATSU, T., WAZAWA, T. & YANAGIDA, T. (1999). Fluorescence resonance energy transfer between single fluorophores attached to a coiled-coil protein in aqueous solution. *Chemical Physics* 247, 163–173.
- JELLEY, E. E. (1936). Spectral absorption and fluorescence of dyes in the molecular state. *Nature* **138**, 1009–1010.
- JIA, K., WAN, Y., XIA, A. D., LI, S. Y., GONG, F. B. & YANG, G. Q. (2007). Characterization of photoinduced isomerization and intersystem crossing of the cyanine dye cy3. *Journal of Physical Chemistry A* **111**, 1593–1597.
- JOO, C., BALCI, H., ISHITSUKA, Y., BURANACHAI, C. & HA, T. (2008). Advances in single-molecule fluorescence methods for molecular biology. *Annual Review of Biochemistry* 77, 51–76.
- KAPANIDIS, A. N., LAURENCE, T. A., LEE, N. K., MARGEAT, E., KONG, X. X. & WEISS, S. (2005a). Alternating-laser excitation of single molecules. *Accounts of Chemical Research* 38, 523–533.
- KAPANIDIS, A. N., MARGEAT, E., HO, S. O., KORTKHONJIA, E., WEISS, S. & EBRIGHT, R. H. (2006). Initial transcription by RNA polymerase proceeds through a DNAscrunching mechanism. *Science* **314**, 1144–1147.
- KAPANIDIS, A. N., MARGEAT, E., LAURENCE, T. A., DOOSE,
 S., HO, S. O., MUKHOPADHYAY, J., KORTKHONJIA, E.,
 MEKLER, V., EBRIGHT, R. H. & WEISS, S. (2005b).
 Retention of transcription initiation factor sigma(70) in transcription elongation: single-molecule analysis.
 Molecular Cell 20, 347–356.
- KHAIRUTDINOV, R. F. & SERPONE, N. (1997). Photophysics of cyanine dyes: subnanosecond relaxation dynamics in monomers, dimers, and H- and J-aggregates in solution. *Journal of Physical Chemistry B* 101, 2602–2610.
- KHIMENKO, V., CHIBISOV, A. K. & GORNER, H. (1997). Effects of alkyl substituents in the polymethine chain on the photoprocesses in thiacarbocyanine dyes. *Journal of Physical Chemistry A* **101**, 7304–7310.
- KIM, H. D., PUGLISI, J. D. & CHU, S. (2007). Fluctuations of transfer RNAs between classical and hybrid states. *Biophysical Journal* 93, 3575–3582.
- KOHN, F., HOFKENS, J., GRONHEID, R., VAN DER AUWERAER, M. & DE SCHRYVER, F. C. (2002). Parameters influencing the on- and off-times in the fluorescence intensity traces of single cyanine dye molecules. *Journal of Physical Chemistry A* **106**, 4808–4814.

- KOLESNIKOV, A. M. & MIKHAILENKO, E. A. (1987). The conformations of polymethine dyes. *Russian Chemical Reviews* 56, 275–287.
- KONING, A. J., LUM, P. Y., WILLIAMS, J. M. & WRIGHT, R. (1993). DiOC6 staining reveals organelle structure and dynamics in living yeast-cells. *Cell Motility and the Cytoskeleton* 25, 111–128.
- KOOPMANS, W. J. A., BREHM, A., LOGIE, C., SCHMIDT, T. & VAN NOORT, J. (2007). Single-pair fret microscopy reveals mononucleosome dynamics. *Journal of Fluorescence* 17, 785–795.
- KORPPITOMMOLA, J. E. I., HAKKARAINEN, A., HUKKA, T. & SUBBI, J. (1991). An isomerization reaction of a cyanine dye in normal-alcohols – microscopic friction and an excited-state barrier crossing. *Journal of Physical Chemistry* 95, 8482–8491.
- KUZMIN, V. A. & DARMANYAN, A. P. (1978). Study of sterically hindered short-lived isomers of polymethine dyes by laser photolysis. *Chemical Physics Letters* 54, 159–163.
- LESOINE, J. F., HOLMBERG, B., MALONEY, P., WANG, X., NOVOTNY, L. & KNAUF, P. A. (2006). Development of an spfret method to measure structure changes in ion exchange proteins. *Acta Physiologica* 187, 141–147.
- LI, J., CHEN, P., HU, X. J., ZHENG, D. S., OKASAKI, T. & HAYAMI, M. (1996). Study on the photofading of nearinfrared absorbing indolenine cyanines dyes. *Chinese Chemical Letters* 7, 1121–1124.
- LUO, G., WANG, M., KONIGSBERG, W. H. & XIE, X. S. (2007). Single-molecule and ensemble fluorescence assays for a functionally important conformational change in t7 DNA polymerase. *Proceedings of the National Academy* of Sciences of the United States of America **104**, 12610–12615.
- MA, X. M., HUA, J. L., WU, W. J., JIN, Y. H., MENG, F. S., ZHAN, W. H. & TIAN, H. (2008). A high-efficiency cyanine dye for dye-sensitized solar cells. *Tetrahedron* 64, 345–350.
- MADER, O., REINER, K., EGELHAAF, H. J., FISCHER, R. & BROCK, R. (2004). Structure property analysis of pentamethine indocyanine dyes: Identification of a new dye for life science applications. *Bioconjugate Chemistry* 15, 70–78.
- MALICKA, J., GRYCZYNSKI, I., KUSBA, J. & LAKOWICZ, J. R. (2003). Effects of metallic silver island films on resonance energy transfer between n,n'-(dipropyl)-tetra methyl indocarbocyanine (Cy3)- and n,n'-(dipropyl)tetramethyl-indodicarbocyanine (Cy5)-labeled DNA. *Biopolymers* 70, 595–603.
- MARME, N., KNEMEYER, J. P., SAUER, M. & WOLFRUM, J. (2003). Inter- and intramolecular fluorescence quenching of organic dyes by tryptophan. *Bioconjugate Chemistry* 14, 1133–1139.
- MASSEY, M., ALGAR, W. R. & KRULL, U. J. (2006). Fluorescence resonance energy transfer (FRET) for DNA biosensors: FRET pairs and Förster distances for

various dye-DNA conjugates. *Analytica Chimica Acta* 568, 181–189.

- MCKINNEY, S. A., TAN, E., WILSON, T. J., NAHAS, M. K., DECLAIS, A. C., CLEGG, R. M., LILLEY, D. M. J. & HA, T. (2004). Single-molecule studies of DNA and RNA fourway junctions. *Biochemical Society Transactions* 32, 41–45.
- MIKHEIKIN, A. L., ZHUZE, A. L. & ZASEDATELEV, A. S. (2000). Binding of symmetrical cyanine dyes into the DNA minor groove. *Journal of Biomolecular Structure and Dynamics* 18, 59–72.
- MISHRA, A., BEHERA, R. K., BEHERA, P. K., MISHRA, B. K. & BEHERA, G. B. (2000). Cyanines during the 1990s: a review. *Chemical Reviews* 100, 1973–2011.
- MOMICCHIOLI, F., BARALDI, I. & BERTHIER, G. (1988). Theoretical-study of *trans cis* photoisomerism in polymethine cyanines. *Chemical Physics* **123**, 103–112.
- MOREIRA, B. G., YOU, Y., BEHLKE, M. A. & OWCZARZY, R. (2005). Effects of fluorescent dyes, quenchers, and dangling ends on DNA duplex stability. *Biochemical and Biophysical Research Communications* 327, 473–484.
- MUJUMDAR, R. B., ERNST, L. A., MUJUMDAR, S. R., LEWIS, C. J. & WAGGONER, A. S. (1993). Cyanine dye labeling reagents – sulfoindocyanine succinimidyl esters. *Bioconjugate Chemistry* 4, 105–111.
- MURPHY, M. C., RASNIK, I., CHENG, W., LOHMAN, T. M. & HA, T. J. (2004). Probing single-stranded DNA conformational flexibility using fluorescence spectroscopy. *Biophysical Journal* 86, 2530–2537.
- MURPHY, S., SAUERWEIN, B., DRICKAMER, H. G. & SCHUSTER, G. B. (1994). Spectroscopy of cyanine dyes in fluid solution at atmospheric and high-pressure – the effect of viscosity on nonradiative processes. *Journal of Physical Chemistry* **98**, 13476–13480.
- NETZEL, T. L., NAFISI, K., ZHAO, M., LENHARD, J. R. & JOHNSON, I. (1995). Base-content dependence of emission enhancements, quantum yields, and lifetimes for cyanine dyes bound to double-strand DNA: photophysical properties of monomeric and bichromophoric DNA stains. *Journal of Physical Chemistry* **99**, 17936– 17947.
- NORMAN, D. G., GRAINGER, R. J., UHRIN, D. & LILLEY, D. M. J. (2000). Location of cyanine-3 on doublestranded DNA: importance for fluorescence resonance energy transfer studies. *Biochemistry* **39**, 6317–6324.
- PATIL, P. V. & BALLOU, D. P. (2000). The use of protocatechuate dioxygenase for maintaining anaerobic conditions in biochemical experiments. *Analytical Biochemistry* 286, 187–192.
- PONTERINI, G. & MOMICCHIOLI, F. (1991). Trans cis photoisomerization mechanism of carbocyanines – experimental check of theoretical-models. *Chemical Physics* 151, 111–126.
- RANJIT, S., GURUNATHAN, K. & LEVITUS, M. (2009). Photophysics of backbone fluorescent DNA modifications: reducing uncertainties in fret. *Journal of Physical Chemistry B* 113, 7861–7866.

- RASNIK, I., MCKINNEY, S. A. & HA, T. (2006). Nonblinking and longlasting single-molecule fluorescence imaging. *Nature Methods* 3, 891–893.
- RASNIK, I., MYONG, S., CHENG, W., LOHMAN, T. M. & HA, T. (2004). DNA-binding orientation and domain conformation of the *E. coli* rep helicase monomer bound to a partial duplex junction: single-molecule studies of fluorescently labeled enzymes. *Journal of Molecular Biology* 336, 395–408.
- RENIKUNTLA, B. R., ROSE, H. C., ELDO, J., WAGGONER, A. S. & ARMITAGE, B. A. (2004). Improved photostability and fluorescence properties through polyfluorination of a cyanine dye. *Organic Letters* 6, 909–912.
- ROY, R., HOHNG, S. & HA, T. (2008). A practical guide to single-molecule fret. *Nature Methods* 5, 507–516.
- RULLIERE, C. (1976). Laser action and photoisomerization of 3,3'-diethyl oxadicarbocyanine iodide (dodci) – influence of temperature and concentration. *Chemical Physics Letters* **43**, 303–308.
- RUST, M. J., BATES, M. & ZHUANG, X. W. (2006). Subdiffraction-limit imaging by stochastic optical reconstruction microscopy (STORM). *Nature Methods* 3, 793–795.
- RYE, H. S., YUE, S., WEMMER, D. E., QUESADA, M. A., HAUGLAND, R. P., MATHIES, R. A. & GLAZER, A. N. (1992). Stable fluorescent complexes of doublestranded DNA with bis-intercalating asymmetric cyanine dyes – properties and applications. *Nucleic Acids Research* 20, 2803–2812.
- SABANAYAGAM, C. R., EID, J. S. & MELLER, A. (2005a). Long time scale blinking kinetics of cyanine fluorophores conjugated to DNA and its effect on Forster resonance energy transfer. *Journal of Chemical Physics* 123, 224708.
- SABANAYAGAM, C. R., EID, J. S. & MELLER, A. (2005b). Using fluorescence resonance energy transfer to measure distances along individual DNA molecules: Corrections due to nonideal transfer. *Journal of Chemical Physics* **122**, 061103.
- SACKETT, D. L. & WOLFF, J. (1987). Nile red as a polaritysensitive fluorescent-probe of hydrophobic protein surfaces. *Analytical Biochemistry* 167, 228–234.
- SAHYUN, M. R. V., SHARMA, D. K. & SERPONE, N. (1995). Mechanisms of spectral sensitization of silver-halides – role of sensitizing dye complexation. *Journal of Imaging Science and Technology* **39**, 377–385.
- SANBORN, M. E., CONNOLLY, B. K., GURUNATHAN, K. & LEVITUS, M. (2007). Fluorescence properties and photophysics of the sulfoindocyanine Cy3 linked covalently to DNA. *Journal of Physical Chemistry B* 111, 11064– 11074.
- SAUERWEIN, B., MURPHY, S. & SCHUSTER, G. B. (1992). Dynamics of solute motion – photoisomerization shows linear-dependence on solvent mass. *Journal of the American Chemical Society* **114**, 7920–7922.

- SCHOBEL, U., EGELHAAF, H. J., BRECHT, A., OELKRUG, D. & GAUGLITZ, G. (1999). New-donor-acceptor pair for fluorescent immunoassays by energy transfer. *Bioconjugate Chemistry* 10, 1107–1114.
- SEIDEL, C. A. M., SCHULZ, A. & SAUER, M. H. M. (1996). Nucleobase-specific quenching of fluorescent dyes. 1. Nucleobase one-electron redox potentials and their correlation with static and dynamic quenching efficiencies. *Journal of Physical Chemistry* 100, 5541–5553.
- SEIFERT, J. L., CONNOR, R. E., KUSHON, S. A., WANG, M. & ARMITAGE, B. A. (1999). Spontaneous assembly of helical cyanine dye aggregates on DNA nanotemplates. *Journal of the American Chemical Society* **121**, 2987–2995.
- SEYBOLD, P. G., GOUTERMA, M. & CALLIS, J. (1969). Calorimetric photometric and lifetime determinations of fluorescence yields of fluorescein dyes. *Photochemistry* and *Photobiology* 9, 229–242.
- SIBBETT, W., TAYLOR, J. R. & WELFORD, D. (1981). Substituent and environmental-effects on the picosecond lifetimes of the polymethine cyanine dyes. *IEEE Journal of Quantum Electronics* 17, 500–509.
- SIMS, P. J., WAGGONER, A. S., WANG, C. H. & HOFFMAN, J. F. (1974). Studies on mechanism by which cyanine dyes measure membrane-potential in red blood-cells and phosphatidylcholine vesicles. *Biochemistry* 13, 3315– 3330.
- SINGH, M. P., JENNINGS, T. L. & STROUSE, G. F. (2009). Tracking spatial disorder in an optical ruler by timeresolved NSET. *Journal of Physical Chemistry B* 113, 552–558.
- SOUTHWICK, P. L., ERNST, L. A., TAURIELLO, E. W., PARKER, S. R., MUJUMDAR, R. B., MUJUMDAR, S. R., CLEVER, H. A. & WAGGONER, A. S. (1990). Cyanine dye labeling reagents – carboxymethylindocyanine succinimidyl esters. *Cytometry* 11, 418–430.
- STEIGER, R., HEDIGER, H., JUNOD, P., KUHN, H. & MOBIUS, D. (1980). Studies on the mechanisms of spectral sensitization of silver-halides. *Photographic Science and Engineering* 24, 185–195.
- SUNDSTROM, V. & GILLBRO, T. (1982). Viscosity-dependent isomerization yields of some cyanine dyes – a picosecond laser spectroscopy study. *Journal of Physical Chemistry* 86, 1788–1794.
- TERASAKI, M., SONG, J., WONG, J. R., WEISS, M. J. & CHEN, L. B. (1984). Localization of endoplasmic-reticulum in living and glutaraldehyde-fixed cells with fluorescent dyes. *Cell* 38, 101–108.
- TOLOSA, L., MALAK, H., RAOB, G. & LAKOWICZ, J. R. (1997). Optical assay for glucose based on the luminescence decay time of the long wavelength dye cy5 (tm). Sensors and Actuators B–Chemical 45, 93–99.
- TOMISHIGE, M., STUURMAN, N. & VALE, R. (2006). Singlemolecule observations of neck linker conformational changes in the kinesin motor protein. *Nature Structural & Molecular Biology* 13, 887–894.

- TOMSCHIK, M., ZHENG, H. C., VAN HOLDE, K., ZLATANOVA, J. & LEUBA, S. H. (2005). Fast, long-range, reversible conformational fluctuations in nucleosomes revealed by single-pair fluorescence resonance energy transfer. *Proceedings of the National Academy of Sciences of the United States of America* **102**, 3278–3283.
- TORIMURA, M., KURATA, S., YAMADA, K., YOKOMAKU, T., KAMAGATA, Y., KANAGAWA, T. & KURANE, R. (2001). Fluorescence-quenching phenomenon by photoinduced electron transfer between a fluorescent dye and a nucleotide base. *Analytical Sciences* 17, 155–160.
- TURNER, D. C. & BRAND, L. (1968). Quantitative estimation of protein binding site polarity. Fluorescence of n-arylaminonaphthalenesulfonates. *Biochemistry* 7, 3381.
- VALEUR, B. (2001). Molecular Fluorescence: Principles and Applications. Weinheim, Germany: Wiley-VCH.
- VAN DER MEER, B. W. (2002). Kappa-squared: from nuisance to new sense. *Reviews in Molecular Biotechnology* 82, 181–196.
- VOGELSANG, J., KASPER, R., STEINHAUER, C., PERSON, B., HEILEMANN, M., SAUER, M. & TINNEFELD, P. (2008). A reducing and oxidizing system minimizes photobleaching and blinking of fluorescent dyes. *Angewandte Chemie-International Edition* 47, 5465–5469.
- VOGELSANG, J., CORDES, T. & TINNEFELD, P. (2009). Singlemolecule photophysics of oxazines on DNA and its application in a fret switch. *Photochemical and Photobiological Sciences* 8, 486–496.
- WAGGONER, A. (1976). Optical probes of membranepotential. *Journal of Membrane Biology* 27, 317–334.
- WAGGONER, A. S., ERNST, L. A. & BALLOU, B. (2009). Building new fluorescent probes. In *Biomedical Optical Imaging* (eds. J. G. Fujimoto & D. Farkas), pp. 120–131. New York: Oxford University Press.
- WAGGONER, A. S. & MUJUMDAR, R. B. (2000). Rigidized trimethine cyanine dyes. 6133445.
- WALDECK, D. H. & FLEMING, G. R. (1981). Influence of viscosity and temperature on rotational reorientation – anisotropic absorption studies of 3,3'-diethyloxadicarbocyanine iodide. *Journal of Physical Chemistry* 85, 2614–2617.
- WEST, W. & GEDDES, A. L. (1964). Effects of solvents and of solid substrates on visible molecular absorption spectrum of cyanine dyes. *Journal of Physical Chemistry* 68, 837–847.
- WEST, W. & PEARCE, S. (1965). Dimeric state of cyanine dyes. Journal of Physical Chemistry 69, 1894–1903.
- WEST, W., PEARCE, S. & GRUM, F. (1967). Steroisomerism in cyanine dyes-meso-substituted thiacarbocyanines. *Journal of Physical Chemistry* **71**, 1316–1326.

- WESTON, K. D., CARSON, P. J., DEARO, J. A. & BURATTO, S. K. (1999). Single-molecule detection fluorescence of surface-bound species in vacuum. *Chemical Physics Letters* 308, 58–64.
- WHEATLEY, P. J. (1959). The crystallography of some cyanine dyes. 2. The molecular and crystal structure of the ethanol solvate of 3,3'-diethylthiacarbocyanine bromide. *Journal of the Chemical Society* 4096–4100.
- WHITE, S. S., LI, H. T., MARSH, R. J., PIPER, J. D., LEONCZEK, N. D., NICOLAOU, N., BAIN, A. J., YING, L. M. & KLENERMAN, D. (2006). Characterization of a single molecule DNA switch in free solution. *Journal of the American Chemical Society* **128**, 11423–11432.
- WIDENGREN, J. & SCHWILLE, P. (2000). Characterization of photoinduced isomerization and back-isomerization of the cyanine dye Cy5 by fluorescence correlation spectroscopy. *Journal of Physical Chemistry A* **104**, 6416–6428.
- WIDENGREN, J., CHMYROV, A., EGGELING, C., LOFDAHL, P. A. & SEIDEL, C. A. M. (2007). Strategies to improve photostabilities in ultrasensitive fluorescence spectroscopy. *Journal of Physical Chemistry A* **111**, 429–440.
- WU, E. S., JACOBSON, K. & PAPAHADJOPOULOS, D. (1977). Lateral diffusion in phospholipid multibilayers measured by fluorescence recovery after photobleaching. *Biochemistry* 16, 3936–3941.
- YASUDA, R., MASAIKE, T., ADACHI, K., NOJI, H., ITOH, H. & KINOSITA, K. (2003). The ATP-waiting conformation of rotating F-1-ATPase revealed by single-pair fluorescence resonance energy transfer. *Proceedings of the National Academy of Sciences of the United States of America* **100**, 9314–9318.
- YILDIZ, A., FORKEY, J. N., MCKINNEY, S. A., HA, T., GOLDMAN, Y. E. & SELVIN, P. R. (2003). Myosin v walks hand-over-hand: single fluorophore imaging with 1.5nm localization. *Science* **300**, 2061–2065.
- YU, A. C., TOLBERT, C. A., FARROW, D. A. & JONAS, D. M. (2002). Solvatochromism and solvation dynamics of structurally related cyanine dyes. *Journal of Physical Chemistry A* **106**, 9407–9419.
- ZHAO, R. & RUEDA, D. (2009). RNA folding dynamics by single-molecule fluorescence resonance energy transfer. *Methods* 49, 112–117.
- ZHUANG, X. W., BARTLEY, L. E., BABCOCK, H. P., RUSSELL, R., HA, T. J., HERSCHLAG, D. & CHU, S. (2000). A singlemolecule study of RNA catalysis and folding. *Science* 288, 2048.
- ZHUANG, X. W., KIM, H., PEREIRA, M. J. B., BABCOCK, H. P., WALTER, N. G. & CHU, S. (2002). Correlating structural dynamics and function in single ribozyme molecules. *Science* 296, 1473–1476.